# YOU'VE GOT E-MAIL: THE IMPACT OF ELECTRONIC COMMUNICATION TECHNOLOGY ON FACULTY PRODUCTIVITY

by

Sheila Frances Culver Elworthy
Bachelor of Arts (Hons.), Simon Fraser University
Master of Business Administration, University of Western Ontario

# THESIS SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF EDUCATION

In Educational Leadership

In the Faculty of Education

© Sheila Frances Culver Elworthy 2008

SIMON FRASER UNIVERSITY

Summer 2008

All rights reserved. This work may not be reproduced in whole or in part, by photocopy or other means, without permission of the author.

# APPROVAL

Name:	Sheila Frances Elworthy
Degree:	Doctor of Education
Title of Research Project:	YOU'VE GOT E-MAIL: THE IMPACT OF ELECTRONIC COMMUNICATION TECHNOLOGY ON FACULTY PRODUCTIVITY
Examining Committee:	
Chair:	Geoff Madoc-Jones, Assistant Professor
	John LaBrie, Dean, Continuing Studies Senior Supervisor
	Bettina M. Heinz, Director, Royal Roads University
	Richard Smith, Assoc Professor, School of Communication
	Joan Collinge Director Center for Online and Distance Education Internal/External Examiner
	Gilbert Wilkes, Assistant Professor, School of Communication & Culture, Royal Roads University External Examiner
	Date: May 21, 2008



# Declaration of Partial Copyright Licence

The author, whose copyright is declared on the title page of this work, has granted to Simon Fraser University the right to lend this thesis, project or extended essay to users of the Simon Fraser University Library, and to make partial or single copies only for such users or in response to a request from the library of any other university, or other educational institution, on its own behalf or for one of its users.

The author has further granted permission to Simon Fraser University to keep or make a digital copy for use in its circulating collection (currently available to the public at the "Institutional Repository" link of the SFU Library website <www.lib.sfu.ca> at: <a href="http://ir.lib.sfu.ca/handle/1892/112">http://ir.lib.sfu.ca/handle/1892/112</a>) and, without changing the content, to translate the thesis/project or extended essays, if technically possible, to any medium or format for the purpose of preservation of the digital work.

The author has further agreed that permission for multiple copying of this work for scholarly purposes may be granted by either the author or the Dean of Graduate Studies.

It is understood that copying or publication of this work for financial gain shall not be allowed without the author's written permission.

Permission for public performance, or limited permission for private scholarly use, of any multimedia materials forming part of this work, may have been granted by the author. This information may be found on the separately catalogued multimedia material and in the signed Partial Copyright Licence.

While licensing SFU to permit the above uses, the author retains copyright in the thesis, project or extended essays, including the right to change the work for subsequent purposes, including editing and publishing the work in whole or in part, and licensing other parties, as the author may desire.

The original Partial Copyright Licence attesting to these terms, and signed by this author, may be found in the original bound copy of this work, retained in the Simon Fraser University Archive.

Simon Fraser University Library Burnaby, BC, Canada



# STATEMENT OF ETHICS APPROVAL

The author, whose name appears on the title page of this work, has obtained, for the research described in this work, either:

(a) Human research ethics approval from the Simon Fraser University Office of Research Ethics,

or

(b) Advance approval of the animal care protocol from the University Animal Care Committee of Simon Fraser University;

or has conducted the research

(c) as a co-investigator, in a research project approved in advance,

or

(d) as a member of a course approved in advance for minimal risk human research, by the Office of Research Ethics.

A copy of the approval letter has been filed at the Theses Office of the University Library at the time of submission of this thesis or project.

The original application for approval and letter of approval are filed with the relevant offices. Inquiries may be directed to those authorities.

Bennett Library Simon Fraser University Burnaby, BC, Canada

# **ABSTRACT**

This qualitative case study explores the connection between electronic communication technology and faculty productivity. Despite the pervasive use of e-mail and other electronic communication media at educational institutions, few studies investigate the relationship of electronic communication to faculty productivity.

In this study, sixteen faculty members from a public, medium-sized, and instruction-oriented college in Western Canada answered detailed questions about their professional on-line behaviour. Questions explored response times to student on-line enquiries, the appropriateness of using electronic communication for certain tasks and topics, and its overall impact on faculty performance. Suggestions were also solicited for improving application of the technology at the college.

Participants tracked all incoming and outgoing electronic communications for seven consecutive days, recording quantity of communications sent and received, time spent, and actions taken. In addition, instructors rated incoming communications in terms of their work-related relevance.

Results were evaluated, using Anthony Giddens' Structuration Theory as a theoretical framework, and were compared to related studies performed at tertiary institutions. The literature review of studies on electronic communication

at higher-education institutions identified themes of gender, age, time, and communication; all of these themes, along with faculty productivity, are explored in the current study.

The majority of participants in this study feel that electronic communication increases their overall productivity. However, most also mention elements of this technology which decrease their performance, citing factors such as miscommunications, reduced face-to-face contact, and excessive volume and time spent on electronic messaging. Time logs maintained during the study reveal that participants may delete or not read electronic communications relevant to their work, suggesting other avenues of communication may be more effective for dissemination of information. This analysis indicates a need for policy regarding response times and suitable on-line behaviour and recommends further training in effective and appropriate use of electronic communication.

**Keywords:** productivity; faculty; e-mail; technology; communications.

**Subject Terms:** electronic communication technology; faculty productivity; faculty performance; college faculty; Structuration theory; out-of-class communication: e-mail.

To Paul

# **ACKNOWLEDGEMENTS**

I feel very fortunate to have had the guidance and support of my committee throughout this engaging process. Thank you John LaBrie, bettina heinz and Richard Smith for your insightful comments and questions that significantly improved the process and final product.

Thank you to Geoff Madoc-Jones, Joan Collinge, and Gilbert Wilkes for a challenging and thought-provoking oral examination.

I am grateful to the sixteen participants in my study who took the time to share their opinions and practices regarding use of electronic communication technology. Other colleagues helped me shape my study and provided encouragement throughout my journey: Linda Cross, Alec Lee, Josje Andmore, Cath Mack, Leanne Webster, Suzanne Thiessen, and Leelah Dawson.

I am most appreciative of the financial support I received from Camosun College Faculty Association, the CA Education Foundation, and the Culver Educational fund.

Finally, I am grateful to my family. Thank you to my Dad and my twin, Diana, for their amazing editorial input, to my children, Maureen, Greg and Brian for their unfailing faith in my ability, and to my husband, Paul, for his love and encouragement.

# **TABLE OF CONTENTS**

Approval	ii
Abstract	III
Dedication	V
Acknowledgements	vi
Table of Contents	vii
List of Tables	x
Chapter 1: Introduction	1
Problem Statement	
Purpose of Study	
Camosun College – Site of Study	6
Definition of Terms	
Electronic Communication Technology	
Faculty Productivity	
Limitations of Study	
Research Questions	
Organization of the Dissertation	13
Chapter 2: Theoretical Framework and Literature Review	14
Giddens' (1984) Structuration Theory	14
Literature Review	
Communication Issues	18
Time Issues	
Gender and Age Issues	
Faculty Productivity	24
Summary of the Research Findings	25
Chapter 3: Research Methodology	27
Method of Inquiry	27
Case Study Approach	
Assessing the Rigor of Qualitative Research	
Research Procedures	31
Ethics Approval	31
Sample Selection	31
Data Collection	
Time logs	35
Preliminary Interview Process	38
Interview questions	
Demographic questions	

Experience questions	.42
Opinion questions	46
Feeling questions	48
Data Evaluation	
Data evaluation of interview responses	49
Data evaluation of time logs	
Chapter 4: Results	61
Demographic Information	61
Saving electronic communications	
Printing electronic communications	
Electronic communication time	
Office hours	
Response times	
Written and verbal policies	
Communication beyond normal working hours	
English as a Second Language and older students	
Tasks for which electronic communication is used	
Tasks for which electronic communication is not used	83
Productivity Issues	
Definition of Productivity	
Increases to Performance	
Decreases to Performance	
Electronic communication and overall Productivity	
Personal strategies for improving Productivity	
College-wide strategies for improved electronic communication usage	
Time logs	
Volume of e-mail	
Average time spent each day on electronic communications	
Categorization of e-mails by relevance to work	
Changes to behaviour patterns	112
Chapter 5: Literature Review Revisited	115
Communication Issues	115
Time Issues	
Gender and Age Issues	
Productivity	
Conclusion	
Chapter 6: Recommendations & Suggestions for Further Research	126
Strategy	
Policy	
Training	
Suggestions for Further Research	
	. 57

Chapter 7: Researcher's Reflections	
References	138
Appendix A: Recruitment letter to School of Business Faculty	
Appendix B: Time Log	
Appendix C: Interview questions	
Appendix D: Informed Consent Form	
Appendix E: List of Codes	
Appendix F: Tabular results of Interviews and Time logs	

# **LIST OF TABLES**

Table 1. Time spent communicating with students outside of class time	63
Table 2. Modes of communication with students outside of class time	64
Table 3. Modes of communication used with others	65
Table 4. Time spent on total communication outside of class time	66
Table 5. Total communication time spent on communication methods	66
Table 6. Comparison of ECT as per interviews to time logs	67
Table 7. Changes in time spent on ECT from one year ago	69
Table 8. Nature of student electronic communications	71
Table 9. Verbal policies on electronic communication technology	75
Table 10. Tasks for which electronic communication is used	82
Table 11. Tasks for which electronic communication is not used	83
Table 12. Definitions of productivity	
Table 13. Do aspects of ECT increase your productivity?	87
Table 14. Explanations of how ECT increases productivity	87
Table 15. Do aspects of ECT decrease your productivity?	91
Table 16. Explanations of how ECT decreases productivity	92
Table 17. Overall, does ECT make you more or less productive?	.100
Table 18. Strategies employed which improve participants' productivity	.103
Table 19. Strategies suggested for improving ECT at the College	.104
Table 20. Number of electronic communications sent by participant	.106
Table 21. Number of work-related e-mails received during the week	.107
Table 22. Average time spent on work related electronic communications	.108
Table 23. Percentage of irrelevant e-mails on which time is spent	.109
Table 24. Summary of e-mails deleted with no action taken	.110
Table 25. Relevant e-mails deleted with no action taken	.111
Table 26. Comparison of the volume of e-mails received by gender	.114

# **CHAPTER 1: INTRODUCTION**

Late one evening, I happened to open my e-mail account, and found the following message:

Hi Sheila,

I am in your afternoon accounting class. I know the final exam is tomorrow morning, but I have been sick for the past week. Can I write the exam later? Please let me know as soon as possible if I can do this.

Thanks. (Anonymous, personal communication, December 13, 2006)

Many thoughts began rushing through my head, all due to this fairly innocent communication. I felt caught in a dilemma; the student had sent the e-mail at 10:00 pm and it was now 11:30 pm. If I responded to the e-mail, I was possibly condoning or even encouraging students to send me e-mails at the last minute and to expect a reply. I was also aware, however, of this student's need for a response. I felt hampered by the asynchronous, one-way nature of electronic communication technology – I could respond to the student but I didn't have the ability to ask questions and then reply in the same conversation as would be the case if the student had telephoned or visited my office. I also felt anxious in that I was dealing with a work-related issue late at night and wondered if the student would even get my response prior to class time. I could easily have missed this

e-mail and could then have had a student annoyed at me for not answering the e-mail in a timely fashion.

My preoccupation with e-mails from students has led me to think seriously about the extensive role of electronic communication technology in my capacity as a college instructor. While electronic communication technology can greatly facilitate my work, I could be spending more time on work-related tasks because of it, yet am not certain that there is a commensurate improvement in either quality or quantity of work-related output. While the technology allows for responses to student and administrative queries from places other than the office, the ability to be reached at any time may lower performance. Due to electronic communication are faculty members spending more time on administrative tasks generated by e-mail or are they spending more time on student engagement activities and academic activities such as course development or research?

I hear my colleagues saying that the pace of their work has increased — there is a sense that data or answers are expected sooner than in previous years (McInnis, 2002). Certainly electronic technology has had an impact on the turnaround time of communication — it is far quicker to send documents and notes electronically than was previously possible. This ability to respond faster appears to carry a cost. When I started working as a college instructor in 1988, student access to faculty was primarily limited to office hours and telephone calls. Weekends often involved college work but the agenda was driven by my sense of urgency regarding my performance, and not by students. With electronic

communication, questions, requests, and documents can be sent at any time of day or night, so there appears to be no down time (Amey & VanDerLinden, 2003). Is this increased sense of urgency justified by perceived increases in workers' productivity and accessibility, and who is reaping the rewards, teacher or student? Does the student garner the greater benefit at the expense of the instructor or is the flexibility a benefit which academics welcome?

As is the case with the tradition of established office hours, I am not convinced that students should always have unrestricted, one-way electronic access to instructors. Perhaps there should be some controls or filtration element in the degree to which all students can electronically make contact with their instructors. The cavalier nature of the medium creates its own set of frustrations for instructors. Students fire off e-mails with a question, only to be followed by another email saying that they solved the problem. Are students passing some of their thinking on to faculty because of a lack of temporal, geographic or financial constraints? This same technology also allows students to contact their professors from off-campus, potentially reducing the time spent on-site. There may be social and pedagogical benefits from students spending more time physically at school in order to engage in the democratic public space provided by a college or university (Davidson, 2007). However, given the ubiquitous nature of the medium, are there perhaps other methods to achieving these worthwhile ends?

Perhaps electronic communication technology is a natural consequence of changes to the delivery of education. In combination with globalization and

society's shifting expectations of higher education, faculty are being challenged to work more productively (McInnis, 2002). Technological innovations such as on-line courses and the ancillary use of electronic communication technology, utilized to make educational delivery more cost-effective, have had a major impact on faculty performance (Amey & VanDerLinden, 2003; Baldwin, 1998; Beam, Kim, & Voakes, 2003; McInnis, 2002; Tabers, 2002).

McInnis (2002) suggests, however, that use of technology does not necessarily lead faculty "to a deeper level of change in the performance of academic work, that is, the shift to a more reflective approach to teaching as scholarship" (p. 56). In particular, he proposes that e-mail has had a profound impact on faculty's teaching and research:

As students increasingly expect the university to fit with their lives rather than expecting to accommodate themselves to institutional time frames, their assumptions about faculty availability for consultation and comment on their work have changed.....E-mail generates expectations that ready access means instant response. Although not yet well researched, this phenomenon may contribute to the fragmentation of faculty time. Evaluation of "response time" becomes likely, but at the risk of trivializing faculty work. (p. 59)

### **Problem Statement**

Despite the pervasiveness of the medium, there is a lack of examination of electronic communication technology's impact on faculty productivity and performance. As previously mentioned, there are both positive and negative aspects to the encapsulation of electronic communication technology within faculty work. Use of electronic communication technology allows for new

productivity options but may also involve additional time spent dealing with quick response time expectations, for example, or concerns about skill levels adequate to deal with technological changes (Baldwin, 1998; Beam et al., 2003).

Technology is altering the definition of faculty work, with time spent incorporating technology into teaching modalities and into communication regimes (McInnis, 2002). Technology is also perceived and utilized differently, depending upon age, gender and skill levels (Ogan & Chung, 2003; Wang & Cohen, 1998). A technology-intense environment, such as a college, is subject to frequent changes relating to instructional innovations and technological advances (Allison, 2004; McInnis, 2002; Tabers, 2002).

This pace and scale of technological change have a significant impact on faculty's ability to keep up with both technological change as well as performance of their academic duties. The purpose of this study is to document current faculty usage of electronic technology as a communication tool, to determine its impact upon performance, with a view to helping to discover and suggest improvements.

# **Purpose of Study**

In studies regarding technology and communication between students and faculty, there appear to be three gaps:

1. There is little attempt to study the relationship between use of electronic communication technology and faculty performance. Discussion includes preference issues over media options, usefulness of varying types of communication, training, and stress levels. Performance is merely hinted

at with the discussion of time spent on electronic communication technology.

- 2. While studies have explored the issues of types of communication and of clarity of communication, they are not from the point of view of faculty (Atamian & DeMoville, 1998; Duran, Kelly, & Keaten, 2005; Montano & Dillon, 2005). What issues do faculty have when using electronic communication technology and do they prefer and use different modes of delivery for different types of communication?
- 3. Most of the studies reviewed were performed almost a decade ago and use and diversity of technology have changed since then. The scale of usage is also dramatically different. Are the results of these studies still valid or have technologies and their use changed in some manner?

The purpose of this study is to explore the connection between electronic communication technology and faculty performance, using Giddens' Structuration Theory (1984) as a theoretical framework for data analysis. Given the immense changes to faculty work resulting from electronic communication technology, a study may bring awareness to the issue and help guide faculty use of this medium. Strategies that are being used by some faculty to improve their performance may be worth adopting as institution-wide policies and may inform staff development models for faculty.

# Camosun College - Site of Study

The research for this topic will utilize faculty from Camosun College in Victoria, British Columbia. This commuter college has been operating for 35

years and currently services more than 8,000 students in credit and vocational programs and another 10,000 students in part-time continuing education courses each year (MacAskill, 2006a). Over 700 of the students are international students from more than 40 different countries and more than 500 are Aboriginal students from 50 different First Nations (Tinis, 2006). Camosun's mission is to be "a comprehensive educational institution providing our community with access to the knowledge and skills relevant to the future economic and social development of the region" (MacAskill, 2006b). The college primarily delivers university transfer courses satisfying the first two years of degree programs, and one and two year programs culminating in certificates or diplomas. Camosun has transfer agreements with 14 universities; the majority of the university transfer students continue their studies at either the University of Victoria or Royal Roads University, both located nearby.

Camosun has recently been granted approval for a four-year applied Bachelor of Business Administration with three possible majors. As such, it is part of a larger national movement amongst colleges to become more engaged as degree-granting institutions (Powell & Snellman, 2004). The college is organized into five schools, these being Business, Arts and Science, Trades and Technology, Access, and Health and Human Services. Class sizes range from 30 to 40 people and the college is primarily a teaching institution as distinct from a primarily research-oriented university.

Camosun employs 860 individuals, 430 of which are instructors belonging to the *Camosun College Faculty Association* (CCFA) (Camosun College, 2008).

The CCFA operates in association with the Federation of Post Secondary Educators, an organization which represents faculty at public colleges in British Columbia. The average age of the student body is 22, many of whom work and attend school at the same time (MacAskill, 2006a). Consequently, there are increasing numbers of on-line, evening, and weekend courses to accommodate the needs of both full-time and part-time students.

### **Definition of Terms**

### **Electronic Communication Technology**

While e-mail is the most prevalent form of computer-based electronic communication currently in use, other forms are continually emerging. Facebook was originally established as an Internet networking service for North American university and college students for posting their personal profiles, exchanging messages and joining groups. Since it opened its registration to the general public in the Fall of 2006, Facebook membership has grown to over 24 million, with 500,000 new users each week (Ranson, 2008; Shields, 2007). MySpace, a popular social networking website offering an interactive, user-submitted network of friends, personal profiles, blogs, photos, music and videos, has attracted more than 70 million users, many between the ages of 14 and 20 (Ray, 2008). Other such social networking sites have also emerged in the market place as the popularity of these sites continues to grow. College students make up a considerable proportion of the rapidly growing use of instant messaging or text messaging for communication (Flanagin, 2005). Some colleges are already using alternate media, such as MySpace, text messaging via student's cell

phones, and web portals, to communicate with their students, as they are finding that college e-mail newsletters are treated as spam by students and often not read (Carnavale, 2006).

Given the emergence of new technologies, the study will not be limited to e-mail exclusively, however it is expected to be the most commonly used format by the Camosun College participants in this study. Thus, electronic communication tools will include e-mail, text messaging, blogs, and websites, with a focus on their utilization for communication rather than as instructional devices. For the purpose of this study telephones and facsimile machines will be excluded. This paper will concentrate on the issues of using digital communication devices in general, so that the findings may be relevant to new emergent formats.

### **Faculty Productivity**

A key variable of the study is faculty productivity, which can be difficult to both define and measure (Antony & Raveling, 1998; Bock, 1997; Massy & Wilger, 1995). Furthermore, faculty may resist settling upon a definition of productivity due to concern that standardized evaluative measures have a political motivation and may shift control over the academic work process from faculty to management (Polster & Newson, 1998).

As this is not a study per se of faculty "output" (such as scholarly papers, teaching loads, research grants, etc.) faculty productivity will be examined as it relates to the faculty member's perceived performance. This analysis will allow for a focus on the communication medium at question, without being sidetracked

with the larger issue of what constitutes academic production and how to measure such output.

Faculty work involves teaching students, research and scholarly activity, committee work, and often community service (Middaugh, 2001). An economic definition of productivity is a measure of production efficiency; a ratio between output and input (McBeath, 1974). Another definition of faculty productivity is an increase in educational outcomes, such as number of students taught or improved instruction, relative to costs, or a reduction in costs used to produce a given set of educational outcomes (Levin, 1991).

According to Middaugh (2001), faculty productivity includes both quantitative and qualitative factors, and while it is easier to establish benchmarks for quantitative elements it is also possible to measure the qualitative aspects. Tierney (1999) contends that, due to the rapid change to post-secondary institutions resulting from technological advancement, new definitions of faculty productivity are needed. The advent of virtual universities, distance learning, electronic publishing options, and changes to course delivery options has dramatically altered the framework for faculty roles (McInnis, 2002; Tierney, 1999). Tierney argues that standard input and output measures are not useful in encouraging an environment of trust between academics and administration, an essential component in an intellectual organization. Instead, the discussion about productivity needs to start from a different premise, one that recognizes that individuals are unique and that everyone has a productive place in the

modern world. Specific measures, then, need to be arrived at by each faculty member, in consult with their department, for definition of their own productivity:

Productivity in an organizational culture is more concerned with ensuring that everyone is playing to the best of his or her ability and that all individuals are operating from the same plan, agreeing about the texture that they desire, and how they will interact with one another. Different musicians have different roles; a group's responsibilities are to one another, to the music, and to the audience. Optimally, the music is joyful and the musician's work is personally fulfilling and appreciated. (Tierney, 1999, p.120)

Productivity may include number of students and sections, number of courses taught, research projects, course development, community service, and administrative tasks performed. Consistent with Tierney's view, faculty members will self-report on whether or not they are able to be more or less productive with the use of communicative technology.

This study will examine a faculty member's perception of their own productivity through their perceived performance enhancements. Consistent with Tierney's view stated above, the study seeks to see if faculty are playing to the best of his or her ability in serving student's needs.

# **Limitations of Study**

This study will not investigate electronic communication technology in relation to on-line course delivery because the offsite nature of such courses makes electronic communication the essential mode of contact. This reality would necessarily skew the results unfairly. Instead, the paper will explore faculty-student communication in on-site courses where other forms of interface, such as written, face-to-face, and telephone, are also available.

Technology has had an enormous impact on teaching, with extensive changes to course delivery through on-line learning, Powerpoint and class management systems such as Blackboard, WebCT and websites. Rather than looking at such well-researched topics as the impact of technological change on teaching, faculty interviews will be directed toward use of technology for communication. Many of the studies reviewed look at the connection of technology and faculty stress, but connecting stress to faculty productivity is beyond the scope of this study.

#### **Research Questions**

The over-arching question to be addressed in this study is: In the perception of college faculty, what impact does electronic communication technology have on their performance? The study is concerned as much with faculty views of electronic communication technology as with the reality of its impact on their productivity. It will examine electronic communication between faculty and their students and electronic communication between faculty and colleagues, administrators and external individuals and organizations.

Sub-questions include:

1. What do faculty members identify as the major issues of using electronic communication? Responses should identify patterns of frequency, magnitude, and timing of hours spent communicating with students and others, and whether use of electronic communication technology has altered time spent on educational and other matters.

- 2. What do faculty members consider timely feedback and what impact does providing timely feedback have on their workload? The literature mentions the significance of timely feedback (Amey & VanDerLinden, 2003; Atamian & DeMoville, 1998; Duran et al., 2005). It is important to get a measure of what faculty perceive to be timely feedback and to see how that characteristic affects their view of their performance.
- Are there methods and techniques that improve the efficacy of electronic communication as a productivity tool? This question will lend itself to identifying methods and strategies that could be used to improve upon job productivity.

# **Organization of the Dissertation**

This dissertation is organized into seven chapters. Chapter 1 provides an overview of the research problem and questions, the potential significance, and limitations. Chapter 2 introduces Giddens (1984, 1991, 1993) as a theoretical framework for this study, and organizes the literature into four themes, being communication, time, gender and age, and faculty productivity. Chapter 3 discusses the qualitative approach to the research and the specific procedures undertaken. Chapter 4 contains the results of the interviews and time logs. Chapter 5 provides an interpretation of the results, including a comparison between this study and previous studies. Chapter 6 discusses recommendations and suggestions for further research into this area. Chapter 7 encompasses the researcher's reflections on the research journey.

# CHAPTER 2: THEORETICAL FRAMEWORK AND LITERATURE REVIEW

# Giddens' (1984) Structuration Theory

Giddens' (1984) Structuration Theory helps explain where electronic communication technology fits into an organization's framework and how it may be used within this framework. Structuration Theory posits that both structure and agency exist in a recursive relationship, where structure constitutes the codes or norms for social action while agency delineates the actions of individuals within the organization (Giddens, 1984). Giddens' theory differs from those which take a more hermeneutic or phenomenological approach of suggesting that society is formed primarily by its individuals, as distinct from functionalist and structuralist theories which suggest that structure moulds the behaviour of individuals (Macintosh, 1994). Instead, Giddens (1984) contends that structure and agency exist in a recursive relationship.

### **Duality of Structure**

The duality of structure notion proposes that agents are influenced by the structure and can also produce or reproduce the structure (Macintosh, 1994). This duality means that workplace culture will be an important factor in the usage and style of electronic communication among agents (Waldvogel, 2007). In addition, agents are purposive and reflexive in monitoring their behaviours within the social structure (Giddens, 1993). This axiom suggests that while the norms of

the College, both written and unwritten, influence the communication that takes place, the agents may make changes to this structure through their behaviour.

Giddens submits that individual acts can bring about social change, but that the social structure will only change when the new methods are institutionalized and form part of the social structure. That phenomenon may be occurring in the usage of electronic communication technology. Some electronic communications received from students are viewed by faculty as too informal, with casual salutations that faculty interpret as disrespectful (Duran et al., 2005). It is possible that faculty are deriving incorrect conclusions about student intent by assuming that the rules that apply to written communications such as letters are appropriate for electronic communications. Perhaps the structure of the electronic communication is evolving and will result in these changes becoming institutionalized as more individuals communicate in the new, more informal format, as hypothesized by Giddens (1984).

#### **Dialectic Control**

The concept of Dialectic Control is another significant component of Structuration Theory, suggesting that dependent relationships are a two-way issue. While superiors have resources giving them power over subordinates, subordinates also have resources giving them influence over their superiors. In a classroom setting, the instructor has attributes for controlling classroom behaviour but students may choose to comply or not with the instructor's demands. Electronic communication is a two-way, but time delayed, interaction between persons in a dependent relationship, such as administrators, faculty,

and students. A student can exert influence simply by sending an electronic communication technology to an instructor which requires a response. The two-way, asynchronous aspect of this communication method is a factor to be considered in analyzing interactions which may occur (Chen, 2001; Montano & Dillon, 2005; Skovholt & Svennevig, 2006).

# The Role of Information in the Knowledge Economy

Giddens' (1991) concept of modernity proposes that individuals have more choices due to the *disembedding mechanisms* afforded by industrialization. The post-industrial era has increased individuals' choices, leading to a desire for more information. Reflexivity is intensified in this *post-traditional* world:

Modernity's reflexivity refers to the susceptibility of most aspects of social activity, and material relations with nature, to chronic revision in the light of new information or knowledge. Such information or knowledge is not incidental to modern institutions, but constitutive of them — a complicated phenomenon, because many possibilities of reflection about reflexivity exist in modern social conditions. (Giddens, 1991, p.20)

Giddens' notion of modernity is consistent with what some are calling the *information society* or the *knowledge economy* with the increasing spread and significance of information in many aspects of life (Morgan, 1998; Webster, 2002). This new economy can be defined as a shift from a post-industrial society to one where production and services are based more on intellectual capabilities and information than on physical inputs or natural resources (Powell & Snellman, 2004). The handling of information, such as e-mail, constitutes an essential part of daily operations and consumes much capital and labour, and could impact faculty workload.

Organizational structures and practices impact the productivity realized from new technologies (Hilton, 2006; Powell & Snellman, 2004). The increasingly volatile global economy may require new structures that allow for the integration of new information with existing expertise (Morgan, 1998). While the organizational models of universities and colleges are shifting, the changes tend to be in response to economic forces and may not improve the handling of information (Fisher & Rubenson, 1998).

# **Summary of Theory**

Communication within and amongst organizations is increasingly occurring via electronic communication technology. As articulated by Giddens (1984, 1991, 1993), use of these media is a function of the social norms of an organization and impacted by the individuals sending and receiving the electronic messages. The reflexive nature of human beings suggests that both electronic communication and the organizational structure, involving rules and resources, may be evolving as use of this relatively new technology matures. This study of the impact of electronic communication technology on faculty performance will involve consideration of both formal and informal policies and procedures impacting communication and use of electronic communication technology within the site of study.

### **Literature Review**

Literature on the use of electronic communication by university and college faculty members can be grouped into three areas of interest, these being

communication, time, and gender and age issues. Literature addressing faculty productivity explores factors contributing to output, including electronic communication, and the conundrum of its definition. These themes are summarized below.

#### **Communication Issues**

As a fairly new medium which potentially reduces or replaces face-to-face and telephone communication, e-mail's ability to convey meaning and information between faculty and their students has been extensively investigated (Atamian & DeMoville, 1998; Chen, 2001; Duran et al., 2005; Flanagin, 2005; Haworth, 1999; Kelly, Keaten, & Finch, 2004; Krause, Hartley, James, & McInnis, 2005). When two professors designed a course in which all contact with instructors was via e-mail, students were generally satisfied with course delivery and instructor availability (Atamian & DeMoville, 1998). One issue that arose, however, was clarity of communication, with half of the students claiming that e-mail messages were confusing. The authors are uncertain whether this mixed response is due to the instructors' style or is an inherent characteristic of e-mail communication.

Conflicting views exist on the ability of electronic communication technology to convey complex tasks. Daft and Lengel's study (1986) claims that the *richness* of the media used for communication determines its capacity for resolving ambiguity and facilitating understanding. Information richness is defined by Daft and Lengel as the ability of information to change understanding over time. Rich media, such as face-to-face or verbal communication, are

personal and interactive while *lean* media, such as e-mail or memos, are impersonal and rely on rules and procedures. Ishii's (2005) study of the use of e-mail for handling equivocal tasks contradicts the findings of Daft and Lengel's (1986) study. Ishii's (2005) survey of university administrators found that e-mail could be used for equivocal tasks but that its success rate depended upon the experience level of the users.

Montano and Dillon's (2005) study explores the use of different technologies for different types of communication within an organization. Their results suggest that, while technologies can strengthen relationships within an organization, certain technologies are better suited for certain types of communications. Telephones, for example, are useful when individuals want to interact, electronic communication technology is seen as convenient, and while websites are valuable for the dissemination of information, they do not foster a sense of belonging to an institution. Despite face-to-face and telephone allowing for communication that is more individualized and personal, there are drawbacks, such as inconvenience (difficulties in arranging meetings) and inequity (students that feel socially excluded by others felt excluded in face-to-face and telephone conversations). Montano and Dillon (2005) propose that technologies such as e-mail create "a fair, homogeneous environment" and may strengthen the individual-to-organization relationship (p.238).

E-mail use may also vary depending upon the size and culture of the academic institution – students at a smaller university used e-mail to make excuses or explain about missed classes, while students at a larger university

used it primarily to enquire about their grades (Duran et al., 2005). It will be interesting to see if this finding is replicated in my study of a medium-sized college. Reticent students, who prefer electronic communication over office visits, may afford instructors increased opportunities for communication (Kelly et al., 2004). E-mail, however, may not always be the most appropriate channel, and does not advance the student's ability to communicate orally (Duran et al., 2005; Kelly et al., 2004). Instructor comments regarding electronic communications include: "Very impersonal but very efficient"; "Quick but can't replace social interaction"; "It is both a blessing and a curse. A blessing when it opens doors of communication. A curse because it takes a lot of time to respond to them." (Duran et al., 2005, p.170). While deemed efficient for procedural information, electronic communication technology may not be appropriate for indepth discussions or for sensitive topics.

Another communication issue concerns the on-line behaviour of students. In Duran et al.'s (2005) study, professors commented that students' writing style was "too informal and sometimes inappropriate" (p.174). The medium of electronic communication lends itself to brevity, particularly with some of the newer approaches, such as on-line messaging (Flanagin, 2005). While one might assume that e-mails should take the form of a letter, that sentiment does not appear to be universally shared and is likely to vary relative to the age of the sender or recipient.

Chen's (2001) study of e-mails between American and Taiwanese students and their American professors revealed divergent, culturally-influenced,

strategies. The Taiwanese student e-mails, for example, tended to be more deferential and, consistent with Chinese culture, used a 'gift-giving strategy' of including compliments to their professors. Chen's study supports the premise that language is socio-culturally determined, leading to the different communication styles of the e-mails.

Waldvogel's (2007) study, however, suggests that workplace culture has a greater influence on styles of communication than socio-cultural factors.

Waldvogel compared e-mail communications within a multinational manufacturing plant which had European New Zealanders, Maori New Zealanders, Pacific Islanders, and other non-English speaking workers to e-mails within a New Zealand educational institution where staff were mainly middle-class, tertiary educated, and mono-cultural European New Zealanders. The study found more open and positive relationships between staff and management at the manufacturing plant, despite the disparities in language, education levels, and social backgrounds, suggesting that organizational culture was more influential than social-cultural factors.

#### Time Issues

A common concern among faculty is that electronic communication technology tends to add to their workload, and so Haworth (1999) set out to determine whether e-mail increased the volume of student-faculty interaction. His study suggests that contact does not increase but is instead redistributed to an alternative form. Rather than augmenting their face-to-face contact with professors, students used e-mail as a substitute. Additionally, Haworth (1999)

found that students with prior internet experience were more likely to use e-mail for contacting professors, and that e-mail increased when professors provided a course-dedicated webpage. This finding is consistent with Ishii's (2005) premise that experienced e-mail users were more comfortable with its use for equivocal tasks.

Duran et al. (2005) propose that the relationship is more complex than presented by Haworth (1999); the extent to which e-mail is augmenting or replacing other communication forms may be determined by "instructor and student variables" (p.171). In addition, their study found a gender difference relating to time, with female faculty receiving significantly more "e-mails from students than their male counterparts" (p.171). They also recommend further research into the variables involved in e-mail usage by faculty and students (Duran et al., 2005).

A longitudinal study on the changes since 1994 of the attitudes and experiences of first year Australian university students suggests that use of email will increase as students spend less time on campus and more time working at a job:

The last decade has seen full-time students progressively spending fewer days on average on campus and reduced hours in class per week (17.6 hours per week in 1994 compared with 15.9 hours per week a decade later). This trend is accompanied by a significant rise in the proportion of full-time students committed to paid employment (47 percent in 1994 compared with 55 percent in 2004). (Krause et al., 2005, p.5)

The authors found a significant inverse relationship between student use of e-mail for contacting staff and peers and time spent on campus, suggesting

that, as students increase their time at paid work, the volume of e-mails to faculty may also increase (Krause et al., 2005). Office hours may no longer adequately meet these students' needs and use of electronic communication may be the replacement (Haworth, 1999). In Atamian and DeMoville's (1998) model of replacing office hours with e-mail, timely feedback to student queries was felt to be essential, requiring instructors to check their e-mail frequently. The results of a survey conducted in 2006 indicate that School of Business students at Camosun College work an average of twenty hours per week in addition to taking three or four courses (Lee, 2006). It shall be interesting to see whether the results of this study validate these previous findings.

#### Gender and Age Issues

Other studies have set out to examine whether age or gender affects faculty use of computers and computer-based technologies in teaching and research (Ogan & Chung, 2003; Wang & Cohen, 1998). Ogan and Chung (2003) determined that women are not technologically challenged and are using and teaching the same technology-based courses as their male colleagues. In fact, women's "attitudes toward change and the use of technology are even more positive than that of the men in the study" (p.367). Female faculty, however, experience significantly more stress than men, concerning both technological factors, such as training and out-dated equipment, and personal stressors, such as health, family and lack of personal time. Wang and Cohen (1998) found "no difference in the use of e-mail between the male and female faculty" but felt an overall conclusion on gender was not possible given their survey imbalance, with

twice as many male participants as female (p.459). Additionally, Wang and Cohen (1998) could not determine "the relationship between age and use of internet services" due to the low number of survey respondents in the older age group (p.459). These inconclusive findings suggest that age and gender should be taken into consideration in any survey of technological issues regarding faculty.

## **Faculty Productivity**

Studies of faculty productivity reveal a difficulty in settling on a common definition, suggesting that it is no longer reasonable or accurate to use traditional measures, such as research production, for evaluative purposes (Antony & Raveling, 1998; Bock, 1997; Massy & Wilger, 1995). Instead, productivity could reflect the range of activities that students benefit from, which include advising, teaching, research and community involvement (Bock, 1997; Katula & Doody, 1990). A frequent recommendation for improving productivity is to increase technological knowledge and many of the studies conclude that faculty development and staff training are essential for effective use of technology (Amey & VanDerLinden, 2003; Brown, Benson, & Uhde, 2004; Shepherd, 2004; Wang & Cohen, 1998).

Cohen's (1996) study of faculty at 26 U.S. universities and colleges found a positive correlation between frequency of use of computer-mediated communication and volume of publications produced. While the study does not prove that computer-mediated communication caused the increase in publications, faculty believed it was beneficial, and that e-mail and network

access offered new tools for research, more timely access to information, and enhanced contact with faculty at other institutions. Presumably, the same premise could be extended to the relationship between faculty and student. The study's measure of productivity was limited to quantity of publications, ignoring quality of production and other facets of faculty performance, such as instruction.

The enhanced ability to communicate with off-campus faculty suggests that new communication patterns are emerging to augment more conventional ones such as conferences, telephone, and correspondence (Baldwin, 1998; Cohen, 1996; Di Petta, 1998). Assignments, for instance, may now be submitted electronically from an offsite location rather than physically delivered, and discussions between parties can be conducted from a distance. This evolution is consistent with Giddens' (1984) notion of Structuration Theory. Giddens' contention that structure and agency operate in a recursive relationship suggests that emergent communication technologies, such as e-mail, will inform communication patterns and individuals will adjust their behaviour accordingly.

#### **Summary of the Research Findings**

Use of electronic communication technology at educational institutions has been increasing and rapidly evolving, generating academic discussion of its role in higher education. As proposed in Giddens' (1984) Structuration Theory, changes in communication strategies between faculty, students, and other parties, have altered scholarly work. Electronic communication's efficacy may be dependent upon the skill level, age, culture, and media-preferences of the participants. A broader definition of faculty productivity may be required than that

of simply measuring scholarly output. Consequently, perceived faculty performance which is inclusive of both traditional measurements of faculty productivity and other aspects of faculty work such as student advising, coaching and mentorship will be used in this study.

# **CHAPTER 3: RESEARCH METHODOLOGY**

# **Method of Inquiry**

Qualitative research in education and other social science disciplines is increasing in usage and importance as a method of inquiry (Gall, Gall, & Borg, 2003; Marshall & Rossman, 2006). This research tradition is appropriate for studying complex social phenomena with the intent of improving understanding and perhaps introducing change. A qualitative study is concerned with context and process and is usually interested in questions of how (Patton, 1987; Toma, 2006). This study of faculty productivity is situated within the context of an instruction-oriented college in Western Canada with all the peculiarities resulting from its unique history and organizational culture. A qualitative approach is appropriate for a study interested in presenting the "broad, panoramic views rather than micro-analyses. The more complex, interactive, and encompassing the narrative, the better the qualitative study" (Creswell, 2003, p.182). While job productivity and performance can be quantified to a certain degree, it contains many qualitative elements, which depend upon the individual faculty member's perception and unique history. This study will explore faculty perceptions as much as the reality of their experience with electronic communication technology and a qualitative study allows for both of these aspects to be captured.

# Case Study Approach

An instrumental case study is an appropriate method for examining innovations with the intention of setting policies or improving decision making (Stake, 2005; Toma, 2006). This methodology also allows for a phenomenon to be studied in-depth within its natural setting and from the perspectives of the participants (Gall et al., 2003; Patton, 1987). It is likely that a multiplicity of views will exist among faculty and a case study approach should capture the nuances of input expected from participants. Given the rapid technological changes involved, the use of a bounded system will capture the relevancy of this case to this particular timeframe and this particular population and will include factors distinctive to this site, such as informal and formal policies and procedures regarding usage of electronic communication technology and other communication strategies. There are many factors involved, such as age of participants, skill levels, attitudes, education, and possibly gender, and a case study allows for those issues to be considered in the analysis.

A major strength of the case study methodology is the opportunity to use multiple sources of data collection (Yin, 1994). While the interview will generate faculty perceptions regarding usage of electronic communication technology, a comparison with actual usage will be attempted. Faculty will be asked to maintain time logs for a one-week portion of the fall 2007 semester.

Demographic information will also be collected, including age, gender, and number of years teaching in academic institutions.

# **Assessing the Rigor of Qualitative Research**

In quantitative research, methods of testing the validity and reliability of study results are well established. Parallel standards can be applied to qualitative research in order to demonstrate rigor (Toma, 2006). These standards of credibility, transferability, dependability, and confirmability will be discussed in terms of this study.

#### Credibility

The credibility of the study depends upon the degree to which the findings reflect the reality from the participants' perspectives and make sense to the readers of the study. Any study performed will be inherently subjective, due to what Toma (2006) refers to as the use of the researcher as the collector and analyzer of data. Drawing on Creswell's (2003) list of key variables to what he terms as *validity*, the following procedures will help affirm the credibility of this research:

- Triangulation strategies that will be used to augment relevance and reliability of the data collected will include the collection of time logs to augment the interviews, use of methods similar to methods employed in other related studies, and a comparison of perspectives on the data with other theories articulated in the literature.
- 2. Transcribed documents from the interviews will be shared with the interviewees to confirm accuracy.
- 3. Rich, thick description will be used to describe the findings in order to convey to the reader the setting within which the case study resides.
- 4. The bias of the researcher will be clearly stated.
- 5. Negative or discrepant information will be disclosed in order to enhance the credibility of the study to the reader.

6. The researcher will spend six months researching this topic and, as an instructor at the college over a twenty-year period, has developed an indepth understanding of the site.

#### **Transferability**

A qualitative study's transferability or generalizability to other populations or settings may be difficult but is still possible (Patton, 1987; Stake, 2005; Thomas, 2006). While the results of this study may be applicable to other colleges, universities and organizations, the size and culture of each organization impacts its use of technology and technological change is perceived differently by different faculties within educational institutions (Shepherd, 2004; Tytherleigh, Webb, Cooper, & Ricketts, 2005; Waldvogel, 2007). Furthermore, the results of this study are applicable to the use of technology in 2007. Even so, transferability will be encouraged by the provision of adequate thick, rich description and contextual information to allow the reader to determine the appropriateness of comparability (Stake, 2005; Toma, 2006). Transferability will be furthered by the inclusion of faculty from departments outside of the School of Business at the College.

#### Dependability

In qualitative research, it is possible that the research design may change over the course of the research project as the researcher becomes more informed during data collection (Toma, 2006). Dependability will be enhanced in this study by full description of the research design, initial approach and any changes that were made during the research process.

#### Confirmability

Even though some subjectivity is inherent in qualitative research, it is important to determine that the study's findings are as free from bias as possible (Marshall & Rossman, 2006; Patton, 1987; Toma, 2006). Confirmability will be strengthened by provision of an audit trail of the research findings, consideration of alternative conclusions, and an awareness of the researcher's own biases.

#### **Research Procedures**

#### **Ethics Approval**

Following the ethics review process, I received permission to proceed with my research from the Research Ethics Board of Camosun College on July 12, 2007 and from the Office of Research Ethics of Simon Fraser University on July 29, 2007.

# **Sample Selection**

I selected faculty from the School of Business that showed a typical range of cases defining the population of interest as well as some faculty at the extremes (Stake, 2005; Thomas, 2006). I included individuals that represented a range on a number of criteria, such as age, gender, area of expertise or discipline, and seniority. The sample was limited to full-time, continuing faculty as part-time faculty have other issues of availability to students that change their approach to electronic communication. This purposive sample involved faculty whose primary role is instruction within the School of Business at Camosun College. I wanted to concentrate on the issues central to teaching rather than the issues of communication that would pertain to those fulfilling an

administrative position. Nine interviews were conducted as I felt that number was sufficient to give representative coverage of the range and the extremes. I was prepared to increase the sample should the interview results show great variety in answers but that wasn't necessary as patterns of response were apparent.

My study also included a subset of faculty members from other Schools within the College and I used similar criteria as mentioned above for selection of seven individuals, resulting in eight male and eight female participants in total. The purpose of the subset was to determine whether experiences noted in the School of Business participants are shared by other faculty. Finding commonalities would help to confirm the results of the study and suggest that transferability to other populations may be appropriate. Emergence of differences would suggest the need for further investigation in a later study.

#### Recruitment of participants from the School of Business

I recruited participants from the School of Business by asking for volunteers at a school-wide meeting attended by about 75 faculty members held in August 2007. I explained that I was completing a doctorate in Educational Leadership through Simon Fraser University and that my dissertation topic was the impact of electronic communication technology on faculty productivity. I told the audience that I needed volunteers for my study and that volunteers would be asked to complete a time log for a one-week period, capturing their use of electronic communication technology. As well, each volunteer would be interviewed one-on-one and the interview would take about one hour. I also

mentioned that the results of the study would be presented in a manner that assured confidentiality and that I would not disclose the names of those who participated in my study. I distributed a recruitment letter (Appendix A) and asked that individuals who were interested or who had questions fill in the form.

Of the fourteen volunteers who completed my form, three ticked the box for "I have some questions. Please call me". One of these three individuals had not yet started teaching and since I wanted persons who could determine whether use of technology had changed from a year ago, I called them to thank them for their offer but to say they didn't meet my criteria for volunteers. When I called the other two potential volunteers, they indicated that they had no questions and were interested in being in the study – apparently they had ticked the wrong box. I rejected three more from the initial fourteen candidates as they did not meet my criteria of teaching full-time while not holding an administrative role in the College. The ten remaining participants fortuitously offered a good range for many of my criteria, such as age, gender, discipline of instruction, and number of years teaching. One of the ten remaining individuals completed the one-week time log but was not interviewed and subsequently dropped out of the study due to health issues, leaving me with nine volunteers from the School of Business faculty.

The interview process included a review of the logs with occasional adjustments being made due to this discussion; I did not have an opportunity to review the log with the individual who was not interviewed. Furthermore, there was no way to compare this participant's answers to interview questions

regarding time spent on electronic communication with the time recorded in the log and so I decided to exclude their log from the study results.

#### Recruitment of participants from other Schools

The faculty members from other Schools within the College were found via personal networking. I asked a few knowledgeable co-workers to recommend participants of various ages, length of teaching career, and disciplines. The twelve recommendations resulted in ten individuals who met my criterion of teaching full-time. Two of the ten possible candidates declined due to time constraints. I wanted all participants from all Schools to complete the time log during the same week but I was not able to arrange a time to explain the log to one individual before that week started and so I ended up interviewing the remaining seven faculty members.

It is worth noting a discrepancy in the selection of participants from the two populations, being the School of Business and the other Schools. The School of Business participants volunteered after having heard my short description of my topic; there was no personal solicitation and no coercion involved. The participants from the other schools were solicited by me over the telephone and had been pre-selected by another individual. I did give a similar description of my field of interest during the telephone call, so they were also aware that I was studying the impact of electronic communication technology on faculty productivity and they were also informed of the research methodology I planned to use. However, they were engaged in the discussion in a format that individuals listening to my talk were not in that I required an answer from them.

did not mention the name of the person who recommended they be included in my study and I made a point of stating that no one at the College would be aware of whether or not they agreed to be interviewed and that confidentiality was assured.

#### **Data Collection**

### Time logs

In addition to being interviewed, the participants completed a time log for seven consecutive days (Appendix B). Completion of the log required them to record their daily electronic communication activities, including time spent, number of e-mails sent and received, and tracking what they did with the e-mails (i.e. replied and/or forwarded, read and deleted, deleted without reading, took other action, or took no action).

Before distributing the time logs to participants I conducted a focus group with three faculty members and an administrator from the School of Business that were exempted from my study due to their administrative roles. The purpose of the focus group was to review my interview questions and my time log. This preliminary enquiry was to ensure that my questions were not leading or biased, that I hadn't missed any major topics, and that the time log was understandable. Based upon the focus group's feedback, I modified the time log to better support the research questions and to make it easier for participants to understand, and changed the wording of some interview questions (Appendix C). None of the results from this preliminary work were included in the study results.

The purpose of the time log was many-fold. I wanted to get a sense of the range and actual volume of electronic communications experienced, and if volume had any connection with their interview comments. I wanted to see if participants' answers to my questions regarding time spent on electronic communication was consistent with their log recordings. I wanted to compare logs from the School of Business participants with those from other schools to determine whether they were similar or different. I also felt that completing the time log would force participants to pay attention to their electronic communication behaviour. This raised awareness might help them give more accurate answers to my probing questions. Participants were requested to return completed logs to my office through inter-office mail. Most of the logs were returned to me in this manner and the remainder were given to me at the time of the interview.

Once participants agreed to be interviewed, I arranged via e-mail or telephone to meet with them individually for about 15 minutes. These meetings all took place between September 26 and Oct. 5<sup>th</sup>, 2007. During that meeting, I gave them a package of seven daily time logs and discussed their completion and the date I wanted them to start record-keeping. The instructions reviewed verbally with each participant were also written on the covering letter to the time log package. I told them that interview and time log results would be kept confidential and asked if they had any questions regarding my study or any parts of their involvement. I answered any questions and mentioned that I would send an e-mail on October 9, 2007 as a reminder to start filling in the time logs. I also

arranged follow-up interview times commencing in the week following completion of the log. I asked that they phone me with any questions and supplied them with three phone numbers. I asked each participant if they wanted any additional telephone or e-mail reminders during the week of completing the log and they all declined. I printed the logs on brightly coloured legal-size paper in the hope that they would stand out on the individuals' desks and serve as a reminder to complete them. One log was printed on brightly coloured letter-size paper, at the request of the volunteer.

One participant telephoned me during the morning of October 9, 2007, the first day of the time log portion of my research. My time log did not have a category for e-mails that were read and then archived. I decided to include these e-mails in the category identified as "replied and/or forwarded" and sent an e-mail out to all participants to explain this change to the time log. Other than my initial reminder e-mail to start the logs, that was the only additional e-mail that I sent during the week in which the time log data was being collected. The logs were completed from Tuesday, October 9, 2007 to Monday, October 15, 2007. wanted to pick a week that was fairly average in terms of anticipated e-mail volume and so I chose to avoid September as the start of term can result in additional volume from both students and administration. I felt that a week in either October or November would tend to be fairly representative of e-mail traffic, both in terms of quantity and quality of communications. The Canadian holiday for Thanksgiving was Monday, October 8, 2007 and I wanted to avoid email traffic from a holiday weekend so we started the next day. I was also hoping to avoid mid-term exams as I suspected that students would be sending more electronic communications shortly before an exam. Some of the faculty involved in my study teach on a quarterly system and some teach on a semester system and some courses have one midterm, others have two, and some have none. Avoiding the midterm exam period was not entirely successful given the variety of teaching schedules and programs involved. During the discussion of the log, I asked participants whether they felt that the week was fairly typical in terms of volume and whether they had changed their behaviour while completing the log. I felt that those questions would give me some sense of whether or not the week selected was fairly normal for them in terms of electronic communication content and whether their behaviour was normal as well.

The interviews were all completed between October 22, 2007 and November 2, 2007. I wanted to interview participants shortly after they had completed their logs so they would be fresh in their minds and would be able to recall the week involved. Each participant selected an interview time that suited their schedules and chose to be interviewed in their office or my office. One person shared an office and so the interview was conducted in the staff lounge, which was empty at the time.

# **Preliminary Interview Process**

Using what Patton (1987) refers to as the interview guide, I chose a conversational approach. In order to encourage divulgence of information on related topics, my wording of questions was not always exact. So that my results

were comparable, all questions were asked somewhere in each interview, though not always in the same order or with the same wording.

In order to inform my interview questions, I interviewed a colleague unofficially regarding the impact of electronic communication upon her work performance. I selected someone who had volunteered to be part of my study but did not meet my criteria of teaching full-time without an administrative role at the college. That discussion identified some additional issues pertinent to my topic and helped me formulate my initial set of interview questions. I subsequently tested out my interview questions by conducting a formal, taped interview with a faculty member who was soon to retire from the School of Business. Our follow-up discussion on both the questions and the process overall led me to modify and rearrange the questions into themes. As previously mentioned, these questions were further modified to incorporate changes suggested by the focus group.

All participants signed an informed consent form prior to the interviews being conducted (Appendix D). The interviews were tape-recorded and took between 30 minutes and one hour each. The tape recorder stopped working during one interview but I didn't notice until the end of the interview. The person graciously agreed to meet one week later and the second half of the interview was taped again. I was concerned that the chance to ponder my questions for the one-week period between interviews might make a difference to that person's subsequent responses since they had had the benefit of hearing all the questions. I took handwritten notes of every interview in addition to the taping

and my comparison of the two second halves of the interview did not reveal any major differences in comments.

My interviews initiated with some numerical questions regarding percentage of time spent on different aspects of electronic communication with different groups. These questions proved challenging for some faculty who were less numerically inclined and so I always asked them first to get them out of the way – there was no real flow of discussion regarding those questions. Once those questions were handled, I allowed the interviewee to digress on further questions, ensuring that all of the questions were eventually answered but not necessarily in the order that I had pre-arranged on paper. A number of the interviewees did talk about critical issues not directly resulting from any of my questions and these discussions are included in later chapters.

The interviews were transcribed during November and the transcriptions were sent to each participant for verification of accuracy. All participants chose to review the transcriptions that were sent to them electronically. Other than a few word changes, all participants said that the transcriptions reflected the interviews accurately. Any suggested word changes were made to the transcriptions.

Qualitative research tends to be emergent rather than pre-ordained, suggesting that questions will be altered as more is learned from participants during the interview process (Creswell, 2003). My review of some of the completed time logs led me to add a final question to my interview questions regarding old e-mails. As well, responses from participants to some of my

interview questions encouraged me to ask questions additional to those identified in Appendix C, but for the most part, I covered all the questions in the interviews.

During the interview, the log results were discussed to ensure that I understood their markings on the log and to verify that they had followed my instructions. In two cases, adjustments had to be made. In one case, my review identified that the person had spent a much larger amount of time on e-mail at home than I had seen in other logs. Our discussion revealed that time spent on non-Camosun work had inadvertently been included. The log was adjusted to remove time spent on those e-mails. Another individual recorded that the e-mails had been sent from the work-site on Sunday, Oct. 14, 2007 when the e-mails had actually been generated from their home. The log was changed to shift those notations over one column to show that they were sent from a remote location.

#### Interview questions

My questions could be grouped into three main categories, being questions about the person's demographics, their experience, and their opinions (Patton, 1987). While I did not explicitly ask any feeling questions, participants expressed their feelings about the subject matter in response to some of my questions. In this section I will explain my rationale for including these questions in the interview and what information I hoped to glean from them.

#### **Demographic questions**

- 1. How long have you been teaching at Camosun College? How long in total have you been teaching?
- 2. Which School and which discipline do you primarily teach in? Which age range are you within? Do you have internet access at home? Is it high-speed internet?

The purpose of these questions was to help locate the participants in relation to others and to potentially allow for transferability of the study findings to other populations. Other studies have found that age and gender have an impact on usage of electronic communication and I wanted to determine whether or not these factors would impact this study (Baldwin, 1998; Ogan & Chung, 2003).

#### **Experience questions**

- 3. What modes of communication are you using for communicating with your students outside of the classroom? (Website postings, blogging, e-mail, face-to-face, telephone, msn)
- 4. What percentage of your working hours is spent communicating via electronic communication (exclude telephone and fax) with students? (0-20%, 20-40%, 40-60%, more than 60%)
- 5. Outside of class time, what percentage of your communication with students, is via: Electronic communication technology \_\_\_Face-to-face \_\_\_Telephone \_\_\_ Other \_\_\_\_
- 6. During an average week, how much time in total do you spend communicating with students outside of class time via all media?
- 7. What percentage of your working hours is spent communicating via electronic communication (exclude telephone and fax) with others, such as administrators, publishers, support staff, etc.? (0-20%, 20-40%, 40-60%, more than 60%)
- 8. What percentage of your communication with others, is via:

  Electronic communication technology \_\_\_\_Face-to-face \_\_\_\_Telephone \_\_\_\_
  Other \_\_\_\_
- 9. During an average week, how much time in total do you spend communicating with administrators, publishers, support staff via all media?

This initial group of experience or behaviour questions was to allow me to get a sense of the volume of electronic communication that faculty were experiencing, both with their students and with other parties, and what modes of communication they were utilizing. This information may aid in the ability of others to assess transferability of this study's results to other populations of interest.

- 10. Do you save e-mails? If so, what percentage of e-mails do you save? For how long? Why do you save them?
- 11. Do you print e-mails? If so, what percentage of e-mails you receive do you print? Why do you print them?
- 12. Has the amount of time you spend on electronic communication changed as compared to one year ago?
- 13. Are there periods throughout the semester when time spent on electronic communication increases or decreases? What factors might be causing this change?

This next set of questions was of lesser importance, but I felt that I would be remiss in passing up the opportunity to ask them of my group. I was curious to see if there was any connection between saving and printing behaviour and participants' age or gender. I also was interested in seeing if there is any trend in terms of usage of electronic communication that is being experienced. I limited my trend analysis to only one year as three of my subjects had only been teaching at Camosun for one year. The final question was designed to give information regarding volume, to determine whether or not it was constant across the term. I was hoping to capture time log data from a fairly normal week and I thought this question might help to confirm whether or not I was successful.

 What is the nature of the student electronic communications? (Asking for assistance on course content, asking for assistance on assignments, administrative (due date, sickness))

This question is similar to one from a survey distributed to faculty at two American educational institutions, one being a small private university and the other being a mid-sized public university (Duran et al., 2005). Faculty overall felt the main reason for students to send an e-mail was excuses, followed by course-related contact and then concern about grades. I wanted to see if similar results

would be found at Camosun College, a college of similar size to the public university (enrolment of 10,500) in the study.

15. Does electronic communication contact with students impact student use of your office hours? If so, how?

An Australian study (Krause, 2005) suggested that electronic communication usage increased as students spent reduced amounts of time on campus. I suspect that office hours may be utilized less by students and I wanted to see if that were true.

- 17. How quickly do you tend to respond to student electronic communication requests?
- 18. Do you have any policy on electronic communication in your course outline or that you discuss with students at the start of a semester?

Atamian and Demoville's (1998) study suggested that timely feedback to student queries was an essential component of successful use of the electronic medium. I wanted to see what range of response times exist and to determine whether or not faculty expectations are shared with students, either verbally or in writing.

- 19. Do you check electronic communication on week-ends? How often do you check? Do you respond to electronic communications on weekends?
- 20. Do you check electronic communication in the evening? How often do you check? Do you respond to electronic communications in the evening?
- 21. Do you check electronic communication when you are on holiday? How often do you check? Do you respond to electronic communications when on holiday?

These questions are related to my interest in expected response time.

Again, I wanted to determine what range existed for responding to electronic communication on weekday evenings, weekends, and while on holidays. The

definition of holidays, which was shared with interviewees, included vacations but excluded statutory holidays, such as Thanksgiving, that occurred during the term of instruction.

24. In terms of work, what typically do you use electronic communication for? In terms of work, what, typically, don't you use electronic communication for?

Many of the studies that formed part of my literature review discussed the ability or inability of electronic communication to deal with certain forms of communication, such as equivocal tasks (Daft & Lengel, 1986; Daft, Lengel, & Trevino, 1987; Ishii, 2005; Roberson, 2004). I wanted to determine whether or not Camosun faculty avoided using electronic communication for certain tasks. I was careful to not give any examples of sorts of communication that may be avoided electronically as I feel quite strongly that electronic communication is a lean media, subject to misinterpretation and miscommunication. I did not want to show my bias to my participants.

28. What strategies have you found helpful for improving your productivity regarding use of electronic communication devices?

I feel that a beneficial outcome of this study may be suggestions for ways to improve performance regarding use of electronic communication and so I asked participants for their personal strategies.

- 31. Did you discover anything while completing the time logs? If so, what?
- 32. Did you change your pattern of dealing with electronic communication while completing the time log? If so, how?
- 33. I notice that many people did not deal with old e-mails on a daily basis. What happens to your old e-mails? How many e-mails do you have in your inbox right now? How many are from yesterday or earlier? Is that the norm for you?

My final group of behavioural questions involved a discussion of the time logs. I was curious whether the process of completing the time log had led to any surprises or discoveries about their electronic patterns of behaviour. I also wanted to document whether behaviours had been adjusted during completion and what the change involved. While studies have shown that behaviours may adjust because of being studied, I felt that explicit behaviour changes may have also occurred (Roethlisberger & Dickson, 1939). As mentioned earlier, the question regarding old e-mails stemmed from a review of some of the time logs prior to starting the interviews. I noticed that very few old e-mails were being dealt with during the seven days. I was starting to suspect that e-mails were handled either immediately or not at all, and I wanted to verify that thought.

### **Opinion questions**

I think that an important component of electronic communication is people's opinions or attitudes regarding its usefulness. I therefore asked for views on expected response times, communicating with different student groups, impact of electronic communication on faculty productivity, productivity definitions, and strategies for improving use of this technology. I will discuss the implications of each question below.

16. What do you consider to be an appropriate response time to student electronic communication requests?

As mentioned earlier, I asked faculty how quickly they tended to respond to student electronic communication requests in Question 17. Question 16 asks for their opinion on what they think is a reasonable policy. One of the concerns

that started me on this investigation was expectations of response time to electronic communication requests sent by students and others, and I was curious to see if a pattern would emerge that may inform policy-setting in this arena.

- 22. How, if at all, is electronic communication with 'English as a Second Language' students different from electronic communication with other students?
- 23. How, if at all, is electronic communication with 'older' students different from electronic communication with other students?

I wanted to determine whether my study would support Chen's (2001) findings of a difference in e-mail communication styles between American and Taiwanese students and to further that study by seeing if any difference was perceived between older students and other students. I also asked faculty to define 'older' in terms of our students.

- 25. How would you personally define productivity with regard to your work as a faculty member?
- 26. Are there certain aspects of electronic communication technology that increase your productivity? If so, please explain.
- 27. Are there certain aspects of electronic communication technology that decrease your productivity? If so, please explain.

My literature review revealed that faculty productivity has not been consistently defined and so I asked faculty to give me their personal definition of productivity (Antony & Raveling, 1998; Bock, 1997; Massy & Wilger, 1995). The question was met with many blank stares when asked, and so I asked more informally: "If somebody said, 'have you had a productive year?' what would you be looking at, in terms of your work at the College?" I then asked for aspects of electronic communication that increased and decreased their productivity.

- 29. If you could change anything about electronic communications, what change would you make that might increase your productivity? What could be done to improve use of this technology at the College?
- 30. Overall, do you think electronic communication technology makes you more or less productive? Why? Is there anything else I should know about the connection between electronic communication technology and your productivity?

Questions 29 and 30 were designed to identify problems that participants are having, or that they perceive, with use of technology at the College, and to generate suggestions for improvements. The final question in this section was to get an overall opinion on the usefulness of electronic communication in terms of productivity and to determine whether there were any other aspects of electronic communication that my questions had not unveiled, by asking if there was anything additional the person wanted to tell me.

# Feeling questions

While none of the questions were designed to be feeling questions, feelings were expressed in response to some of the interview questions. Some respondents were rather ambiguous in their answers and so I asked them outright whether or not they liked electronic communication as an additional question. Feeling responses were also fairly standard in response to the question regarding what participants felt electronic communication should not be used for and what elements of electronic communication decreased their productivity. Many of these sentiments are captured and expressed in the quotations in later chapters.

#### **Data Evaluation**

My initial evaluation of the data generated by the interviews was a qualitative analysis whereby I coded all the transcriptions. I coded the data at least three times, with each pass refining and reducing the number of codes so that I ended up with a manageable number of themes (Appendix E). These themes and their implications are discussed in Chapter 4.

# Data evaluation of interview responses

The second method of evaluation was to compare the responses to each question between the School of Business faculty and the faculty from the other schools ("other" faculty). I reviewed the interviews many times, particularly as faculty members occasionally gave answers to different questions in response to another question asked. This iteration helped me to gain familiarity with the data and to find common themes (Creswell, 2003). Tables of responses or information that I want to emphasize in my discussion are included in the relevant sections of Chapter 4 and the remainder of the tables are presented in Appendix F. The following is a description of the evaluation method used for each of the 33 questions asked in the interviews.

Question 1: How long have you been teaching at Camosun College? How long in total have you been teaching?

Question 6: During an average week, how much time in total do you spend communicating with students outside of class time via all media?

Question 9: During an average week, how much time in total do you spend communicating with administrators, publishers, support staff, etc. via all media?

Questions 1, 6 and 9 generated a numeric reply that is reported for all 16 participants; School of Business faculty, and other faculty. I also combined the

answers to Questions 6 and 9, allowing me to report on the total time spent on all work-related communications outside of the classroom.

Question 2: Which School and which discipline do you primarily teach in? Which age range are you within? Do you have internet access at home? Is it high-speed internet?

While I collected data on the disciplines within which each instructor taught, I omitted that data from the results and only reported the number of individuals from each of the four Schools in the study. Some of the disciplines have few faculty members and I was concerned that disclosing disciplines might make some of the participants' comments identifiable. The age range and gender of participants was recorded by each of the three groups. The high-speed internet question generated a yes/no response, and again, this information was tabulated for the three groups. The individual who was temporarily without a home computer was included in the 'no' category for this question but answered the remaining questions which involved a home computer (i.e. "do you respond to e-mails in the evening?") based upon their prior behaviour. The lack of a computer meant that the time log for this person showed no time logged in the evenings or on weekends.

Question 3: What modes of communication are you using for communicating with your students outside of the classroom? (Website postings, blogging, e-mail, face-to-face, telephone, msn)

During the interview process, I read out a list of possible modes for communicating with students and participants identified all of the methods that they typically employ. I reported on the number of individuals using each method within the three groupings. As no participants were using MSN, I excluded it

from the tabular results. I combined 'blogging' and 'other' as only one individual is using blogging as a communication medium.

Question 4: What percentage of your working hours is spent communicating via electronic communication (exclude telephone and fax) with students? (0-20%, 20-40%, 40-60%, more than 60%)

Question 7: What percentage of your working hours is spent communicating via electronic communication (exclude telephone and fax) with others, such as administrators, publishers, support staff, etc.? (0-20%, 20-40%, 40-60%, more than 60%)

Participants chose one of four categories that represented their average amount of time spent on this activity.

Question 5: Outside of class time, what percentage of your communication with students is via: electronic communication technology, face-to-face, telephone, or other?

Question 8: What percentage of your communication with others is via: electronic communication technology, face-to-face, telephone, or other?

I asked each participant to assign 100% to the above-mentioned four categories. A number of participants' first answer did not add up to 100% and so I asked them again to assign values to the categories. In order to identify the number of hours spent on communication via each modality, I cross multiplied the results to Questions 5 and 6 and the results to Questions 8 and 9. I combined these totals and reported on the range of time spent on each modality.

In order to validate the comments made during the interviews regarding time, I compared the total time participants said they spent on electronic communication to the actual time spent during the week, as reported on their time logs. One School of Business instructor said in the interview that they spent 17 hours a week on electronic communication and yet their time log showed 5 hours for the week. The estimate of another School of Business instructor, who

had included instruction time in their estimate of student communication time, was double the amount of time they actually spent during the week of the time log. My follow-up discussion with that individual did not illuminate the portion of time spent teaching and the portion spent on communicating outside of teaching hours. Both instructors commented that the time log occurred during a slow week but, given the large discrepancy between their estimated and actual times, I removed both of their estimates from the data. I found the responses consistent with the time log entries for the remaining individuals.

# Question 10: Do you save e-mails? If so, what percentage of e-mails do you save? For how long? Why do you save them?

The first question in this series of questions generated yes/no answers and I charted those answers by grouping. The answers to what percentage is saved were very vague and so I didn't capture that data in a table. I categorized into four groupings the answers to the "for how long" question. I grouped the 'why' responses, to which I received more than one reason from some people, into common reasons.

# Question 11: Do you print e-mails? If so, what percentage of e-mails do you print? Why do you print them?

Many individuals initially said no in response to this question, but then added "almost never" or "very rarely" to their answer. I grouped the responses into "sometimes", "very rarely" and "never" to reflect answers to the first two questions together. The answers to the "why" portion of the question were organized into six categories, with responses exceeding the number of participants as some instructors gave more than one reason.

Question 12: Has the amount of time you spend on electronic communication changed as compared to one year ago?

The responses were grouped into three categories; time spent has increased, decreased or no change. Reasons for the change were grouped thematically, with reasons for increased time shown separately from explanations for decreased time.

Question 13: Are there periods throughout the semester when time spent on electronic communication increases or decreases? What factors might be causing this change?

The first part of question 13 generated a yes/no response, followed by factors which participants who answered "yes" feel might be causing fluctuations.

Question 14: What is the nature of the student electronic communications? (Asking for assistance on course content, asking for assistance on assignments, administrative (due date, sickness))

Faculty identified the many reasons why students would contact them via electronic communication and these reasons were arranged into six categories, with more than one response given by most participants.

Question 15: Does electronic communication contact with students impact student use of your office hours? If so, how?

When answering this question, faculty identified whether they felt student use of electronic communication increased, decreased or had no impact on usage and so I combined the two questions into one table, reflecting the direction of the change and whether a change was felt. One individual, who has never experienced office hours without electronic communication present, was not prepared to give an opinion and so was excluded from the results.

Question 16: What do you consider to be an appropriate response time to student electronic communication requests?

Question 17: How quickly do you tend to respond to student electronic communication requests?

The answers to these two questions required some interpretation and clarification. I realized that my questions were unclear and so I checked the answers I received with follow-up questions to ensure I understood the response given. If, for example, an interviewee said that they tend to respond within a few hours, I asked: "If a student sent you an e-mail at 10pm would your answer still be "within a few hours"? My follow-up question led to some of them explaining further, which helped me to get a better sense of what they felt was an appropriate response time and what their typical behaviour was. The answers were grouped into three timeframes that reflected the range of answers given.

Question 18: Do you have any policy on electronic communication in your course outline or that you discuss with students at the start of a semester?

I tabulated whether the person had a verbal policy, a written policy or no policy on e-mail that they shared with their students, and then listed both the verbal and written policies concerning electronic communication. The response I received one person with a written policy was unclear, so I verified the information by looking at their course outline.

Question 19: Do you check electronic communication on week-ends? How often do you check? Do you respond to electronic communications on weekends?

Question 20: Do you check electronic communication in the evening? How often do you check? Do you respond to electronic communications in the evening?

Question 21: Do you check electronic communication when you are on holiday? How often do you check? Do you respond to electronic communications when on holiday?

I summarized by faculty grouping the answers given to this series of questions. In order to validate the answers to these questions, I also compared the answers regarding evening and weekend checking to the time logs results.

Question 22: How, if at all, is electronic communication with 'English as a Second Language' students different from electronic communication with other students?

Question 23: How, if at all, is electronic communication with 'older' students different from electronic communication with other students?

The interview responses were clustered into two binary categories; communication is different and communication is not different. For the positive responses, I grouped them thematically and discuss the themes in Chapter 4.

Question 24: In terms of work, what typically do you use electronic communication for? In terms of work, what, typically, don't you use electronic communication for?

The responses to both of these questions were grouped thematically and presented in tables. Many individuals gave more than one answer to each question, particularly to the question of what they would not use electronic communication for.

Question 25: How would you personally define productivity with regard to your work as a faculty member?

The answers to this question were grouped thematically and presented in a table, with more than one answer often expressed by the participants.

Question 26: Are there certain aspects of electronic communication technology that increase your productivity? If so, please explain.

Question 27: Are there certain aspects of electronic communication technology that decrease your productivity? If so, please explain.

Question 30: Overall, do you think electronic communication technology makes you more or less productive? Why? Is there anything else I should know about the connection between electronic communication technology and your productivity?

Responses were grouped into yes/no categories. The multiple perceptions of ways in which electronic communication increased and decreased productivity were grouped into similar categories and presented in tables, with more than one response per person expressed. The explanations given in responses to Question 30 were very similar to the reasons expressed in response to Questions 26 and 27. Consequently, all similar comments made to these three questions were combined.

In order to confirm the person's overall opinion regarding electronic communication, I compared the individual responses to Question 30 to those given to Questions 26 and 27. One person's answer to this question was inconsistent with comments made to prior questions. A review of their transcription revealed that their response was about e-mail specifically and not electronic communication in general. A follow-up conversation clarified that they felt electronic communication was helpful and so I adjusted the tabulated results to reflect this sentiment. The second part of Question 30 was intended to generate any additional comments regarding electronic communication technology or productivity that my questions hadn't elicited. I was told a few interesting stories in response to this question or at the end of the interview. One participant answered 'no' to this question but, after I had turned off my tape

recorder, told an interesting story about a departmental conflict related to e-mail.

I turned the recorder back on and they recounted the story for me. The responses to this question are included throughout my paper.

Question 28: What strategies have you found helpful for improving your productivity regarding use of electronic communication devices?

Question 29: If you could change anything about electronic communications, what change would you make that might increase your productivity? What could be done to improve use of this technology at the College?

Strategies and ideas for change were categorized and presented in a table, grouped into common responses. More than one response was offered by many participants.

#### Question 31: Did you discover anything while completing the time logs? If so, what?

I wanted to fully utilize the value of the participants by asking for their observations regarding the usefulness of completing the time logs. Good research should forward the learning of the participants and I was interested in what value, if any, had resulted from this experiment (Patton, 1987). Discoveries that participants shared with me are discussed in Chapter 4.

# Question 32: Did you change your pattern of dealing with electronic communication while completing the time log? If so, how?

Due to the fact that this aspect of their lives was being studied, It is likely that all participants changed their pattern of dealing with electronic communication to some degree (Roethlisberger & Dickson, 1939). Additionally, I wanted to see if any individuals were conscious of, or had explicitly changed their behaviour in some manner while completing the time log. I tabulated the yes/no

answers to the first question and describe those results, along with the changes in Chapter 4, for those that self-identified as having made a change.

Question 33: I notice that many people did not deal with old e-mails on a daily basis. What happens to your old e-mails? How many e-mails do you have in your inbox right now? How many are from yesterday or earlier? Is that the norm for you?

I added this question after receiving time logs back from participants that showed very little time being spent on old e-mails. I suspect that if an e-mail isn't dealt with immediately, then it is possible it will never be dealt with, and I hoped this additional question would shed some light on this conjecture. I also wanted to see if any pattern exists between behaviour with old e-mails and the number of e-mails in inboxes. Marshall and Rossman (2006) hypothesize that the subjective view is what really matters in qualitative research but that if a researcher wants to make more objectivist assumptions they need to triangulate the interview data with other data. I did want to corroborate the responses to Question 33 – if individuals, for example, responded that they dealt with e-mails immediately, then I thought they should have few e-mails in their inbox. I compared the responses in Question 33 to the time logs. I did not ask the question regarding age of the old e-mails as some of the interviews took place in locations where the individual did not have access to their computer. While they might feel comfortable answering my question on the number of old e-mails in their in-box, I felt it was unrealistic to think they would know the answer to the question about age of old e-mails, suggesting that answers might not be accurate. Some of the choices made in a case study, such as what questions to ask, must consider issues of access and hospitality (Stake, 2005). As these

questions were asked at the end of the interview, I felt I would overstep my welcome by returning to numerical, detailed questions. Furthermore, I could reasonably estimate the age of old e-mails based upon the answer given to the prior question on volume.

# Data evaluation of time logs

Roberson's (2004) study of e-mail as a communication tool included time logs but, as only 7 of the 45 participants completed the logs, the validity of her findings may be limited (p.74). I included time logs in my study in order to see how individuals in my study were managing their electronic communications, and to see if my study would garner results similar to Roberson's (2004). What volumes were they experiencing and how much time were they spending on electronic communication? If they felt a message was relevant, what were they doing with it? What portion of incoming communications is relevant to their work and, if irrelevant, were they still being read?

The time log results were tabulated and responses were compiled into the two faculty groupings. Comparisons were made between the School of Business faculty and other faculty on the number of electronic communications sent by participants, time spent on electronic communication each day, number and handling of electronic communications received each day, handling of old electronic communications, and the percentage of communications that were deleted without reading or on which no action was taken. A comparison was also made of the volume of communications received during the work week and on the weekend by gender. In order to avoid double-counting, electronic

communications received prior to the day are excluded from all tables other than those which specifically look at old e-mails. If a communication is received on Monday, for example, but not dealt with until Tuesday it will show up in both Monday's and Tuesday's volume. The comparative tables are presented in Chapter 4, with complete tabulations of the time logs appearing in Appendix F.

# **CHAPTER 4: RESULTS**

## **Demographic Information**

I interviewed nine individuals from the School of Business and seven from other Schools at Camosun. While the purpose of the study was not to have full representation from all disciplines at Camosun College, I ensured that the School of Business faculty included membership from all of the major areas of study. I intended to focus my study on the School of Business faculty, but wanted to ensure study reliability and transferability and therefore included faculty from other schools at Camosun. The other faculty participants included individuals from three of the four other schools at the College; Trades and Technology, Arts & Science, and Health & Human Services (Table F1 in Appendix F).

The number of years working as a faculty member at Camosun College ranged from 1 to 25 years (Table F2). The overall average was 8 years for the School of Business faculty and 7 years for the other schools. The number of years working as a faculty member at any educational institution ranged from 2 to 27 years (Table F3). The data collected regarding years of service proved reliable across the various schools. Since technological attitudes could vary relative to the age of the individuals involved, I selected participants from different age groups (Wang & Cohen, 1998). Although I interviewed individuals from every decade between ages 20 and 60, the ages are weighted towards the higher ranges, with 75% of the individuals being between the age of 40 and 60

(Table F4). That phenomenon is consistent with the age of individuals teaching at post-secondary institutions – a national study of U.S. colleges and universities found the average age of faculty is between 49 and 51 (Huber, 1998).

Unfortunately, I have no participants from the 61 to 70 years category.

My study included interviews with eight women and eight men, with equal numbers of males being from the School of Business faculty and the other faculty group and with five women interviewed from the School of Business faculty and three women interviewed from other schools. All but one participant has high-speed internet access at home. The aforementioned individual was in the process of moving and so the lack of a computer was temporary but did exist during the time of my study.

The most common methods of communicating with students are e-mail, face-to-face, telephone, and website postings. Blogging and Facebook were also being used for communicating with students and for faculty communications. All but one participant reported spending less than 20% of their working hours on electronic communication with students outside of class time. That person reported spending 25 hours per week on average communicating with students. A lot of these discussions involved one-on-one evaluations of the students' progress and could be considered part of that instructor's teaching load. When this individual is removed from the findings, the School of Business results are very similar to those of the other faculty, with time spent communicating with students outside of class ranging from two to eight hours per week (Table 1).

This data would indicate that there does not appear to be a significant difference between e-mail use amongst various disciplines.

Table 1. Time spent communicating with students outside of class time

	All faculty (15)	SOB faculty (8)	Other faculty (7)
Range	2 to 8 hours	2 to 8 hours	4 to 8 hours
Average	6 hours	6 hours	6 hours

*Note.* One School of Business (SOB) faculty member was omitted from the responses as part of their reported time included teaching time. SOB = School of Business.

A wide range of responses was expressed for the percentage of time spent communicating with students via different modalities, capturing the different approaches to communication being used (Table 2). All faculty interviewed spent at least 20% of their time communicating with students face-toface, but some faculty use this medium almost exclusively, reporting that 98% of their communication is via this modality. The average responses to this question indicate that face-to-face is the favoured method for all faculty members, with School of Business faculty using this method 52% and other faculty using this method 82% of the time. The range for use of electronic communication with students is from 1% to 70%, with the School of Business faculty averaging 38%, more than double the 15% average of other faculty. This relatively higher use of electronic communication could be explained by the larger number of evening courses, where students meet on a weekly basis, offered by the School of Business to accommodate their students, many of whom attend school part-time. Faculty interviewed from other schools reported that many of their students are

attending full-time, cohort-based programs and they are spending a lot of their time on campus. The telephone is used on average less than 10% of the time.

Table 2. Modes of communication with students outside of class time

	All faculty (16)	SOB faculty (9)	Other faculty (7)			
Range of responses						
ECT	1 to 70%	18 to 70%	1 to 45%			
Face-to-face	20 to 98%	20 to 80%	50 to 98%			
Telephone	0 to 33%	0 to 33%	0 to 5%			
Average resp	onse					
ECT	28%	38%	15%			
Face-to-face	65%	52%	82%			
Telephone	7%	9%	3%			

*Note.* Percentage of time spent communicating via different modalities, rounded to the nearest whole percent. ECT = electronic communication technology. SOB = School of Business.

Most of the respondents spend less than 20% of their time communicating electronically with other instructors, administrators, support staff, and others, with the remaining three individuals spending 20 to 40% of their time on these activities. Electronic communication and face-to-face are the most popular modes of choice, with telephone used less than 10% on average (Table 3). While the overall average for all faculty members is almost evenly split between use of electronic and face-to-face communication, School of Business faculty use electronic communication slightly more, being 52% of the time as compared to 41% usage by other faculty in the study. This difference could relate to the fact that School of Business faculty normally teach at least one evening class per week and so tend to vary their hours on campus to accommodate their teaching

schedule. It could also relate to the fact that School of Business faculty are communicating electronically with their students more, due to the aforementioned reasons, and there is a spillover effect to their other communications.

Table 3. Modes of communication used with others

	All faculty (16)	SOB faculty (9)	Other faculty (7)			
Range of responses						
ECT	8 to 80%	15 to 80%	8 to 80%			
Face-to-face	15 to 90%	20 to 80%	15 to 90%			
Telephone	0 to 20%	0 to 20%	0 to 5%			
Average resp	onse	· · · · · · · · · · · · · · · · · · ·				
ECT	47%	52%	41%			
Face-to-face	48%	42%	57%			
Telephone	5%	6%	2%			

*Note.* Percentage of time spent communicating via ECT, face-to-face and telephone, rounded to the nearest whole percent. ECT = electronic communication technology. SOB = School of Business.

The total amount of time instructors said they are spending during an average week communicating with both students and others ranged from 6 to 21 hours, with an average of 13 hours (Table 4). This table omits the answers given by two School of Business faculty members; one individual, as previously mentioned, included teaching time in their answer and the other had a large difference between their actual time and their estimate of time. Based upon the contractual 37.5 hour work week for Camosun faculty, participants are spending about 35% of their time on all communications outside of the classroom (Camosun College, 2008).

Table 4. Time spent on total communication outside of class time

	All faculty (14)	SOB faculty (7)	Other faculty (7)
Range	6 to 21 hours	7 to 18 hours	6 to 21 hours
Average	13 hours	12 hours	14 hours

*Note*. Two School of Business (SOB) faculty members' responses were omitted from the table. Responses indicate time spent on all modes of communication for an average week, rounded to the nearest half hour. SOB = School of Business.

Of that time, faculty members said almost half is spent on electronic communication (Table 5). School of Business faculty profess to spending 6 hours a week on average communicating electronically as compared to other faculty who spend only 4 hours per week on average. A difference also exists between the amount of face-to-face time, with School of Business faculty averaging 5 hours a week in comparison to the 10-hour average for other faculty.

Table 5. Total communication time spent on communication methods

	All faculty (14)	SOB faculty (7)	Other faculty (7)
Range of responses		,	
ECT	½ to 7 hours	2 to 7 hours	½ to 7 hours
Face-to-face	2 to 17 hours	2 to 9 hours	4 to 17 hours
Telephone	0 to 2 hours	0 to 2 hours	0 to 1 hour
Average response			
ECT	5 hours	6 hours	4 hours
Face-to-face	7 hours	5 hours	10 hours
Telephone	1 hour	1 hour	½ hour

*Note.* Two School of Business (SOB) faculty members' responses were omitted from the table. Responses indicate time spent on each mode of communication employed during an average week, rounded to the nearest half hour. ECT = electronic communication technology. SOB = School of Business.

In order to validate the responses received to my questions, I compared the actual amount of time spent on electronic communication from the time logs to the responses given to my interview questions (Table 6). The actual range of time spent was very similar to the time participants said that they spend. It is worth mentioning that participants, having completed the time logs within weeks of the interviews being conducted, could have based their answers to my questions on this prior knowledge. However, to do so would have been difficult as I didn't ask for actual hours but instead asked for percentages of their time spent using different media for communicating. The average time spent on electronic communicating was a bit lower than the time stated in the interviews, with an overall average of 3 hours per time logs as compared to 5 hours from the interviews. Almost half of the participants had mentioned that the volume of electronic communications was lighter than normal during the week captured in the time logs.

Table 6. Comparison of ECT as per interviews to time logs

	All faculty (14)	SOB faculty (7)	Other faculty (7)
Range of ECT time per interviews (Table 5) <sup>a</sup>	½ to 7 hours	2 to 7 hours	½ to 7 hours
Range of ECT time per time logs	1 to 7 hours	1 to 7 hours	1 to 5 hours
Average ECT time per interviews (Table 5) <sup>a</sup>	5 hours	6 hours	4 hours
Average ECT time per time logs	3 hours	4 hours	3 hours

*Note.* Responses indicate time spent on ECT during an average week, rounded to the nearest half hour. ECT = electronic communication technology. SOB = School of Business.

<sup>&</sup>lt;sup>a</sup> Two School of Business (SOB) faculty members' responses were omitted.

## Saving electronic communications

Fourteen of the faculty surveyed save at least some of their electronic communications, which for all of these individuals constitute e-mails. The range of time over which e-mails were saved varied from two terms to indefinitely. The main reasons given for saving e-mails was that faculty thought they might need to refer to the e-mail at some point in the future or they may require the e-mail as proof of a conversation or a decision, and one person said that saving was consistent with their personality, "because I'm a pack rat!" (Subject O, personal communication, November 1, 2007). Equal numbers of men and women saved at least some of their communications. One person said they "like to have the records. If you want to go back and check something, it's there. Quite often there are attachments that I might want to use again" (Subject E, personal communication, October 29, 2007). Another reason given was as evidence of an interaction:

Or as evidence, like if you're communicating with a student, and there is some issue going on, you need to be able to pull up those e-mails and say, "On such-and-such a date, we had this conversation." So almost in a 'CYA' [cover your ass] kind of way, I might save e-mails. (Subject B, personal communication, October 22, 2007)

## Printing electronic communications

Of the twelve individuals who occasionally print e-mails, their reasons pertained mainly to providing a paper trail, or allowing them to take the information to a place where they did not have access to a computer, such as a meeting, or as a reminder "to do something about it" for items that require follow-

up action (Subject N, personal communication, October 31, 2007). "If there is a lot of information in a document or if it's more than a page, I like to read it on a paper" (Subject O, personal communication, November 1, 2007). Three faculty members said that they did print e-mails but "almost never" and one individual professed to never print e-mails (Subject E, personal communication, October 29, 2007).

#### **Electronic communication time**

The amount of time spent on electronic communication had increased from a year ago for 5 faculty members, 7 individuals felt it had decreased, and 4 felt there was no significant change (Table 7). School of Business faculty were evenly divided among the three categories whereas the majority of the other faculty had experienced a decline in time spent.

Table 7. Changes in time spent on ECT from one year ago

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Increased	5	3	2
Decreased	7	3	4
No Change	4	3	1

*Note.* ECT = electronic communication technology. SOB = School of Business.

Of the individuals who had experienced an increase in time spent on electronic communication, 3 had only taught for a few years; one person said that "as I work as a faculty member longer, I have more connection, more staff I need to involve" (Subject F, personal communication, October 31, 2007). The remaining 2 individuals felt the increase was "mostly because of contact with

students. They're very at-home with that system." (Subject M, personal communication, October 26, 2007). "I would say that it's been a very steep curve in the last three to four years, actually. My e-mail traffic goes crazy a day before the exam" (Subject C, personal communication, November 2, 2007).

Three of the faculty members who had experienced a decline in electronic communication usage said it was directly related to a change in their job at Camosun College. A personal choice to consciously reduce their usage of electronic communication was offered as an explanation for 3 other faculty members. One person has "made a conscious decision to scale back and so now I keep it, I think, fairly firmly under control" (Subject K, personal communication, October 23, 2007), and another felt the decline is because they are "not e-mailing as much, so I'm not getting as much back either" (Subject L, personal communication, October 25, 2007). The final person felt their time reduction resulted from an improved ability to communicate electronically.

Of the faculty surveyed, 75% felt that time spent on electronic communication fluctuated during the term. Almost all School of Business faculty members agreed with this statement, while just over half of the other participants agreed. Most individuals felt that electronic communication increased before exams or assignments were due. The two remaining individuals had conflicting explanations, saying that communication increased over time during the term, versus saying that communication was busy at the start of term. The nature of student communications were frequent questions regarding course content or communications regarding absenteeism (Table 8).

A tendency to limit or want to limit electronic communication in general and e-mail usage in particular was mentioned by a number of interviewees.

I don't think e-mail itself actually affects productivity. I think it is the amount of e-mail affects productivity. And I think, what I've seen, and certainly have noticed in the last year or two, that the fact that I'm writing less e-mails and spending less time on e-mail has definitely increased my productivity and improved my quality of life. (Subject L, personal communication, October 25, 2007)

Table 8. Nature of student electronic communications

	All faculty (16)	SOB faculty(9)	Other faculty (7)
Questions regarding course content	11	7	4
Absenteeism	9	4	5
Arranging a face-to-face meeting	2	1	1
Emotional issues/needing contact	2	1	1
Sharing files/jokes	1	0	1
Negotiating assignment extensions	1	1	0

Note. More than one response per person is possible. SOB = School of Business.

### Office hours

Eleven of the faculty felt that student use of electronic communication decreased student use of office hours. "I know students are swamped and it's much less time-consuming to e-mail me than it is to come and see me" (Subject B, personal communication, October 22, 2007). Four said that their office hours were not impacted. Two individuals make a point of meeting with students rather

than responding to their questions on e-mail, saying that "e-mail is just to book the face-to-face with them" (Subject K, personal communication, October 23, 2007) and "I encourage people to actually make an appointment in my office hours, so I can actually see people as opposed to respond to e-mails" (Subject L, personal communication, October 25, 2007). One individual had never taught without e-mail and so had no way of knowing if office hours would be different without this technology.

## Response times

The majority of the 16 respondents indicated that they think a reasonable response to student e-mails is within 24 hours, with four expressing the view that the response should happen within six hours. One person indicated that they didn't want to respond too quickly due to a concern about establishing an expectation on the part of students that would not be easily maintained.

But I try to do my e-mail correspondence with students during my office hour times, just so that they don't get used to me writing a reply at nine-thirty at night, so that they wouldn't expect me to always be able to respond within an hour. So they sort of know what's an expected pattern, and then I can maintain that pattern, as opposed to maybe not being able to maintain checking e-mails seven times a day, in the evenings. (Subject G, personal communication, October 23, 2007)

A relatively new instructor felt that three days was an appropriate response time. Since most classes meet twice a week, this policy allows them,

in most cases, to respond to the student in person, rather than send an electronic response.

So if they haven't heard [from] me, they'll see me before the time is up and they'll ask me the question in person, and so then I don't have to respond to them in e-mail. I can just delete it. So it's beautiful. And that's changed, because in the first semester, I was all about getting a response to them within a day and it's just silly, I think. (Subject P, personal communication, October 30, 2007)

Most participants actually respond to student electronic communications much faster than they had indicated was appropriate, with the majority responding within less than 6 hours and all responding within 24 hours.

I do check my e-mail at night, I do check my e-mail on weekends, I do respond to students if I see their e-mails, I respond immediately. The average turn-around on student e-mails is probably no more than 6 hours and it's probably less than that. (Subject B, personal communication, October 22, 2007)

As long as I have access to a computer, roughly I would check e-mail every one hour. Sometimes students were surprised, "Wow. I get a response right away." (Subject F, personal communication, October 31, 2007)

For some participants their answer varied, depending upon whether they were referring to weekday or weekend responses.

I respond back really quickly because I'm on the system most of the time, at home or over here....If it happens to be

on a weekend and you happen to be out of town, or choosing not to stay connected for the weekend, you may end up having two days before you respond, I tell them. Not the norm. (Subject I, personal communication, November 1, 2007)

Comments made clearly indicated that most people considered it a priority to respond to student enquiries with one person noting that their "faculty response time is probably double to triple" their 24-hour response time to students (Subject C, personal communication, November 2, 2007).

## Written and verbal policies

Only two participants include a written policy on their course outlines regarding electronic communication; one asks students to phone or to send an email if absent and the other says that e-mail is the best mode of communication. Neither instructor discusses response times or whether or not e-mail will be looked at on the weekend or in the evening. Some instructors told students verbally when they could expect a response and one instructor discussed access on the weekend, but most do not mention expected response time in their verbal discussions with students (Table 9). One instructor tells students "if I haven't replied within 24 hours, I didn't get your message" (Subject D, personal communication, October 22, 2007) and another tells them "if they need a quick reply, they should send me an e-mail because the culture at Camosun is not about the telephone, it's about e-mail" (Subject N, personal communication, October 31, 2007).

Table 9. Verbal policies on electronic communication technology

	All faculty (11)	SOB faculty(7)	Other faculty (4)
I will respond withinhours/days	4	2	2
Use e-mail to contact me	3	2	1
Use e-mail anytime	3	2	1
Use e-mail to set up an appointment	2	0	2
I check e-mail on weekends	1	1	0
If I haven't responded in 24 hrs. email me again	1	0	1

Note. Table indicates the verbal policies communicated by the 11 faculty members that explain their electronic communication policy to students. More than one response per person is possible. SOB = School of Business.

One person experienced a very upsetting incident due to a lack of explicit policies regarding electronic communication response time. The administration of their department implicitly, yet unofficially, supported the student's contention that they could expect a response to an electronic communication within less than 24 hours.

There was an exam scheduled. It was a make-up exam, and students were told "You either are at the make-up exam or you fail the course." It was posted on the website; each student who had to do the make-up was sent a personal email. It was announced in class several times. The night before the exam, so I'm talking the exam was on a Friday at 1 pm; at about 7 in the evening on the Thursday night, a student sent me an e-mail saying "I know the exam is tomorrow, but I don't know what time." Now I had an appointment the following morning, so I came onto campus and checked my e-mail just prior to going into the exam. So it's 12:30 and I e-mail the student back and I say "It's in half

an hour." The student didn't show up....So anyway, he argued, "I sent you an e-mail in sufficient time for you to respond to tell me what time this exam was" and my argument was "That is a ridiculous expectation that you would have that kind of turn-around necessarily." ...And his argument was supported by the administration. So, is there an expectation in the administration's eyes? I asked them ... him ... if there was an expectation in the administration's eyes around turn-around time for student e-mails. The official answer was "No" but it was obvious, because the student's argument was supported, that yes there is. The student had the right to expect that I would have responded to that e-mail in enough time for the student to have received it and arrived at campus on time for the exam. (Subject B, personal communication, October 22, 2007)

While this example strongly points to a need for a written policy on electronic communication, it also raises the issue of the ability of responsibility to be transferred, often without agreement or acknowledgement, through this medium (Roberson, 2004).

## Communication beyond normal working hours

Most faculty members interviewed are checking their websites or their email in the evenings, on weekends, and while on holiday. One person felt that working outside of normal hours was a reasonable expectation for faculty.

That doesn't strike me as unreasonable anymore. I used to think that that was an unreasonable expectation but I don't anymore because students are working on their assignments on the weekend and it's of enormous help to them if I can

actually respond over the weekend. And if you think about it in terms of our holiday time and how the flexibility of our work allows us to stagger our hours, I no longer think that's an unreasonable expectation. (Subject H, personal communication, October 24, 2007)

## Checking on weekends

All but one person checks their e-mail on weekends. The individual currently without home internet access responded that their normal behaviour was to check e-mail on weekends, and so they are included in the 'yes' category. The majority of all faculty members check their e-mail once or twice a day. One person checks their e-mail 10 to 15 times per day. "I'm a huge nerd, apparently" (Subject D, personal communication, October 22, 2007). A comparison with the time logs revealed that all but two faculty members had included time on the weekend on their logs for at least one, if not both days, consistent to their verbal answers. One participant was without a home computer during the study and so it was logical that no computer time was recorded for the weekend.

Most responded to communications received over the weekend, partly to reduce the number of e-mails they would have to deal with later:

Even though I try not to respond, I do respond. I can't help it sometimes. Again, I think it comes back to me wanting to deal with it there and then, so I can delete it. So that, the next day, I don't have that e-mail sitting in my inbox. (Subject L, personal communication, October 25, 2007)

Some respond only "if a person's in crisis....I could tell (name) was really upset, and so I responded as soon as I saw this" (Subject P, personal

communication, October 30, 2007). The other reason given for sometimes responding was "if they're going to have an exam on Monday, I will respond. This weekend, I sent out, I think, five bulletin board messages, because students asked good content questions and so, if I worked on it for one, I gave it to all of them". (Subject P, personal communication, October 30, 2007)

# Checking on weekday evenings

Of all faculty members interviewed, 12 check their electronic communication on weekday evenings, with most checking once or twice a night. Two faculty members are less frequent checkers, checking only on evenings prior to student assessments being due. An occasional checker said that "it depends on how late I leave work sometimes. Or if I'm expecting a reply, I might check it, to see what the reply was" (Subject G, personal communication, October 23, 2007). One person said they look for electronic communications a few evenings in a week, when "there's an assignment due" (Subject H, personal communication, October 24, 2007). Most individuals who are checking their incoming messages are willing to respond to those messages. Four faculty members do not check their electronic communication during the evening. One person refuses to check Camosun e-mail on the evening or the weekend, saying that "cut-off time is 5 o'clock" (Subject K, personal communication, October 23, 2007). Instead, that person gives their personal e-mail address to students and tells them to use it "if it's an emergency, you need to know about an assignment; then I'll maybe respond" (Subject K, personal communication, October 23, 2007). Excluding the person who is currently without a home computer, the time log results support the interview comments for all but one individual. That person said in the interview that they did not check their e-mail in the evenings but the time log indicates that they checked four out of the five week day evenings included in the study.

# Checking on holiday

Eleven of those interviewed said that they check their electronic communication when on vacation. Most of those eleven individuals were checking once or twice each week, although two individuals said they check their system every day. A person that checked for new e-mails even when out of the country felt a strong obligation to respond to students — "I have to find an internet connection, to be able to respond" (Subject J, personal communication, October 25, 2007). One respondent has not yet had a holiday at Camosun College and so did not want to predict their behaviour during holiday time in terms of e-mail checking. Responses to e-mail are lower on holiday than on weekends or evenings, with only two-thirds of the eleven 'checkers' saying that they would always respond and the remainder saying that sometimes or rarely they respond.

A common reason for checking while on holiday was to deal with e-mails so as not to feel overwhelmed by the volume when the person arrived back at work.

Yes, it's mostly to get rid of the junk, so you aren't faced with two trillion messages when you get back. At least you can delete them even if you don't feel like responding to everything. You can get rid of the ones you know you don't have to deal with. (Subject M, personal communication, October 26, 2007)

Some were checking e-mail on holiday because the Camosun e-mail address was also used for their personal mail and so they saw Camosun material by default when checking for personal messages.

Only one individual has a policy of never checking work-related electronic communications from home in the evenings, on weekends, nor while on holiday.

I believe that e-mail has sort of an energy with it and that's why I don't open them at home because Camosun I feel, comes into my house and I don't want that in my family life. So I only look at e-mails here, I come up here and open it and then I can respond and react here while I'm getting paid for it. (Subject K, personal communication, October 23, 2007)

## English as a Second Language and older students

One-half of all participants indicate that electronic communications with 'English as a Second Language' (ESL) students are no different from those with their other students. Comments remarked on the difficulty of understanding requests made by the ESL students due to their English skill levels. "I rarely seek clarity from English speakers, but I sometimes have to seek clarity with one more e-mail, to make sure I understand the question" (Subject J, personal communication, October 25, 2007). Two faculty members suggest that this subset of the student population use e-mail less than other students, while another member contradicts this view, stating that ESL students are more

comfortable with e-mail than with face-to-face contact. "Student by student, 'English as a Second Language' speakers prefer e-mail as compared to face-to-face. That's been my experience....I would assume that they're shyer about the contact with the teacher; that would be my assumption" (Subject J, personal communication, October 25, 2007). This observation that ESL students have little contact in any forum was shared by another faculty member:

ESL students, or the international students, which is like an ESL component, don't communicate easily, period, either verbally or in e-mail. They just tend to be invisible in the classroom, and they either make it or they don't. So where they're getting their support unless they come and see me, I have no idea. So I reply to whatever they send me, it's often broken English, but I get what they're saying so I hit the 'reply' and talk back. (Subject N, personal communication, October 31, 2007)

Half of the participants felt that electronic communication with 'older' students differed from communication with other students. The main explanation given is that e-mails from older students "tend to be more clear, they are better at composing their message, they tend to be more polite and they tend to be more appreciative" (Subject B, personal communication, October 22, 2007). The format of writing style was seen as different from younger students, often being more formal. "Still not a formal letter but more formal than what some younger students would write. They might be less likely to say "Hey" or use all lower case letters or that sort of thing" (Subject G, personal communication, October 23, 2007). One person had a hard time understanding younger students, who are

"so used to communicating with their friends where there's no punctuation, there's no capitalization, it's short bursts on messenger [text messaging] that sometimes their questions are not composed very clearly" (Subject I, personal communication, November 1, 2007).

Two interviewees felt that older students "tend not to use e-mails frequently, [they're] much more comfortable with face-to-face" or they phone (Subject D, personal communication, October 22, 2007). An opposing view was offered by another participant, who felt that older students used e-mail more because of their busy lives; "I think because the younger students are here all the time. The older students just whip [home]...they've got family or whatever, they tend to have to be at home more" (Subject O, personal communication, November 1, 2007). Faculty members identified an 'older student' as being over 30 years of age.

# Tasks for which electronic communication is used

Electronic communication is seen by participants as useful for logistical tasks, such as arranging appointments, sending and receiving files, and for communicating with others (Table 10).

Table 10. Tasks for which electronic communication is used

	All faculty (16)	SOB faculty(9)	Other faculty(7)
Logistical arrangements	11	6	5
Communication	7	4	3
Everything	1	0	1

*Note.* More than one answer per person is possible. SOB = School of Business.

### Tasks for which electronic communication is not used

In response to my question asking if there were any tasks that they typically would not use electronic communication for, most faculty members said they would not use it for discussing personal issues or sensitive topics, saying they would "never scold somebody on e-mail" (Subject N, personal communication, October 31, 2007) nor "use e-mail when the issue is controversial or emotional" (Subject B, personal communication, October 22, 2007) (Table 11).

Table 11. Tasks for which electronic communication is not used

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Discussing personal problems, sensitive topics	11	7	4
Discussing lecture material or complex assignments	3	3	0
Department-wide discussions with colleagues	3	0	3
Mass e-mails to students	2	1	1
Nothing – I use it for everything	1	0	1
Sending large files	1	1	0
Responding to angry e-mails	1	1	0

*Note.* More than one answer per person is possible. SOB = School of Business.

A few faculty choose not to discuss lecture material over the internet as they feel it is either a waste of their time to repeat the lecture or too difficult to convey complex issues via this medium. One person said that electronic communication was useful for all work-related tasks and they used it for all communications; "I don't think I'd be nearly as productive, in terms of

communicating with the students, in being able to get the messages out, department-wide or school-wide, there's no way I'd be nearly as productive" (Subject D, personal communication, October 22, 2007).

## **Productivity Issues**

## **Definition of Productivity**

When asked about what constituted productivity for them, faculty members responded with definitions which can be grouped into two major categories of student-related performance measures and of personal or department/college goals (Table 12).

Table 12. Definitions of productivity

	All faculty (16)	SOB faculty(9)	Other faculty(7)
Creating new course material	9	3	6
Helping students to progress	7	3	4
Being accessible to students	4	3	1
Being productive in dept/college	4	2	2
Personal growth/learning in subject	4	2	2
Getting marking done on time	3	1	2
Good collegial relations	1	1	0

*Note.* More than one answer per person is possible. SOB = School of Business.

A discussion of performance in terms of students elicited comments such as "getting my marking done in a timely way" (Subject E, personal communication, October 29, 2007), "making sure that I have really clear material for students" (Subject A, personal communication, October 29, 2007) and being

approachable so that "students can come and talk to me" (Subject L, personal communication, October 25, 2007). Student progress was often mentioned: "So productivity, for me, is when I see students becoming enthused, engaged, being successful in what they're doing, not only academically but practically" (Subject N, personal communication, October 31, 2007). "My internal definition of productivity would be a measurement of how many students were able to progress - and that's a very loose term - in their lives as a result of being here" (Subject B, personal communication, October 22, 2007). The difficulty of monitoring productivity in this environment was also mentioned.

I find productivity in this job excruciatingly difficult to measure because we're doing so many things that are all funnelled and connected to service the students, but I find it really hard to check things off a to-do list on a daily basis.

(Subject J, personal communication, October 25, 2007)

Personal aspects of productiveness included "gains in terms of your understanding of your area", and learning new teaching methods (Subject M, personal communication, October 26, 2007). Departmental measures included working on departmental projects and forming positive collegial relations.

Additional comments mentioned the ability to create new material for courses and for departmental purposes.

Productivity would be hooked in with creativity. So if I am in a good space, I can create and produce an incredible amount in terms of prep, in terms of packages, designing, anything curriculum development just happens, just like that. So it's hooked in very much with that release and, if you're

free to think, then you can create. (Subject K, personal communication, October 23, 2007)

I would look at my effectiveness, and I would sort of be measuring how effective I was able to be that year, in terms of new curriculum, updates, development, embracing new learning teaching kind-of methods and discovery of that nature, as opposed to how many e-mails I answer, how many students I processed, or the ratios of those kind of interactions. And so, if I was not able to do a lot of that kind of creative stuff, then I would say I was not obviously very productive, because I was wasting all my time in non-value-added activities. You know, just shuffling those e-mail lists and stuff like that. (Subject I, personal communication, November 1, 2007)

Productivity definitions may include number of students and sections, number of courses taught, research projects, course development, community service, and administrative tasks performed (Middaugh, 2001; Tierney, 1999). The definitions advanced by Camosun faculty concentrate primarily on student advancement, course development, number of courses taught, and administrative tasks and are consistent with those found in the literature for faculty at a non-research based college (McInnis, 2002).

#### **Increases to Performance**

Thirteen participants felt that certain aspects of electronic communication increased their productivity and the three remaining individuals declared this technology did not enhance their ability to do their work (Table 13).

Table 13. Do aspects of ECT increase your productivity?

	All faculty (16)	SOB faculty(9)	Other faculty (7)
Yes	13	8	5
No	3	1	2

*Note.* ECT = Electronic communication technology. SOB = School of Business.

A valuable function of e-mail is its ability to deal with procedural tasks, such as sending and receiving documents and setting up meeting times (Table 14). Faculty mentioned sending print jobs to the print shop, confirming meetings, answering simple, logistical questions, and sharing resources by e-mail among faculty teaching similar courses.

Table 14. Explanations of how ECT increases productivity

	All faculty(13) <sup>a</sup>	SOB faculty(8)	Other faculty (5)
Good for quick, procedural communications	6	2	4
Asynchronous communication possibilities	6	4	2
Less time-consuming for students to e-mail than visit	5	3	2
Facilitates communication	4	2	2
Saves time - Good for logistical tasks	2	1	1
Reduced need for meetings	1	0	1
Sharing of resources	1	0	1

*Note*. More than one answer per person is possible. ECT = electronic communication technology. SOB = School of Business.

<sup>a</sup>Explanations were offered by the 13 faculty members who answered "yes" to the question shown in Table 13.

If I can send a quick e-mail to audio-visual, that to me is increasing productivity. When I think about it at ten o'clock at night, I can go send an e-mail, and at 7:30 in the morning, that e-mail is responded to. When I'm a couple of days late, because I've worked really hard on creating an exam, I can send a pdf to the print shop. Eight o'clock the next morning, she sends me a confirmation that she received that. (Subject J, personal communication, October 25, 2007)

I really like e-mails for clarification from other staff...."Do you want to go for lunch and have a meeting today?" "Yes, what time?" We can get that done really quickly. We don't have to play telephone tag or anything like that. I think really those are the biggest things. (Subject A, personal communication, October 29, 2007)

Electronic communication allows faculty to respond relatively quickly to student questions, without the need to be in the same place at the same time - "it does save time if they just want a quick answer, especially to a question from class or an assignment that they might be doing" (Subject A, personal communication, October 29, 2007). Communication with other faculty and support staff is also sped up: "if you had to reach someone on the phone every time you needed to communicate with them, you might not be able to get the point across as quickly" (Subject G, personal communication, October 23, 2007).

One comment made was that electronic communication reduced the number of departmental meetings being held as information was being shared on-line among faculty. This person works in a department where instructors eat lunch together most days of the week, a behaviour which may be pivotal to this

strategy working. "We spend a lot of time communicating as a group, so we're kind of together a lot" (Subject O, personal communication, November 1, 2007).

The ability to send files or messages without two people having to be together was seen as an important value offered via electronic communication. This value was especially seen as important as many faculty members are not on campus during regular hours and setting up departmental or private meetings can be challenging; "it means, to communicate with somebody, they don't have to be there at the same time that you have to be there, so the time is very flexible" (Subject O, personal communication, November 1, 2007).

Just think about the days before we had e-mail and we needed to communicate in an organization the size of Camosun. So it's the primary way, even preferable to phone for me because, what I like about e-mail is that I can send a message and people can deal with it in their own sweet time. If this moment is not good for them, I don't have to phone them and interrupt them and play the back-and-forth message thing. (Subject B, personal communication, October 22, 2007)

Electronic communication also appears to facilitate communication with students that may not be known by their instructor otherwise. One instructor mentioned a few incidents where she dealt with students via e-mail that she wouldn't recognize if she saw them in her classroom.

Some of them have some problems. And so we just try to deal with them....It is emotional for them. So we try to deal with the problems over email....But I just sit down and calmly

say "Do this and do this" and then I will usually follow-up with a phone call. Or I will ask them for their phone number or I will say "Please call me to talk about this." But I went through probably two weeks of emails with this one fellow that was having all these problems, and I think maybe that was the only way that he could relate. I don't know that he felt capable of talking to somebody about this....So I just clarified everything in the email for him. (Subject A, personal communication, October 29, 2007)

This observation supports Montano and Dillon's (2005) study findings that suggest electronic communication is a good medium when individuals feel unequal or less powerful in a relationship. Reticent students may find increased opportunities for communication with faculty as other studies also found these students prefer electronic communication over office visits (Kelly et al., 2004). Giddens' (1984) concept of Dialectic Control suggests that dependent relationships are a two-way affair; electronic communication may be enhancing such relationships by giving students a resource which enables them to exercise influence in their interaction with their instructors.

Interestingly, many of these characteristics that have been cited as enhancing faculty performance will also be discussed as items seen as timewasters and limiters of productivity.

### **Decreases to Performance**

Fifteen of the sixteen participants felt that aspects of electronic communication decreased their productivity (Table 15).

Table 15. Do aspects of ECT decrease your productivity?

	All faculty (16)	SOB faculty(9)	Other faculty (7)
Yes	15	9	6
No	1	0	1

*Note.* ECT = Electronic communication technology. SOB = School of Business.

Participants mentioned many more negative factors than positive, and yet, overall, the majority of participants felt that electronic communication greatly enhanced their performance. It is important to remember this phenomenon when reading through the following list. The main reasons for decreased productivity advanced by the fifteen respondents were issues of time spent on too many communications with little value, concerns about reduced face-to-face contact, and potential miscommunications (Table 16).

The most commonly expressed problem with electronic communication was the volume of mail that was received and felt to be of little or no value to the recipient.

When you come in on a Monday morning and there's thirty-five e-mails there, then there's things I have to sort through. So I am guilty of doing work at night and then others are guilty of the same and it causes me to slow down the next morning when I have this screen full of e-mail. That's problematic for me. (Subject J, personal communication, October 25, 2007)

Table 16. Explanations of how ECT decreases productivity

	All faculty (15) <sup>a</sup>	SOB faculty(9)	Other faculty(6)
Too much volume of e-mails	6	5	1
Too much time reading information of little value	5	4	1
Blanket e-mails (mail lists, cc'ing)	3	3	0
Junk e-mail	3	2	1
Reduces face-to-face contact	3	1	2
Slower than face-to-face	3	2	1
E-mails are easily misunderstood and create conflicts	2	0	2
Not a rich enough medium	1	1	0
Technical problems make it unreliable	1	0	1

*Note.* More than one answer per person is possible. ECT = electronic communication technology. SOB = School of Business.

Getting "too many unnecessary e-mails" was also seen as a big waste of time (Subject L, personal communication, October 25, 2007). The interviewee defined 'unnecessary' as being ones that are "non work-related, which aren't necessarily going to have an impact" (Subject L, personal communication, October 25, 2007).

Many people mentioned that they waste time reading e-mails of little value, a behaviour that was identified in my analysis of the time logs as well:

In that 'not related to work' category, not directly related to work or what I'm doing on a day-to-day basis category, I'd like to say that I just delete e-mails, but I'm in the habit of actually reading the darn things and not just deleting them if I

<sup>&</sup>lt;sup>a</sup>Explanations were offered by the 15 faculty members who answered "yes" to the question shown in Table 15.

see they're there, and that does take up some time. (Subject C, personal communication, November 2, 2007)

There are some general business school communications, which you can speed-read very quickly and you sift through that information. Some of it's important immediately, some of it's not. But it takes up time, it just takes up a tremendous amount of time, and having too many of those, I think can reduce productivity. (Subject L, personal communication, October 25, 2007)

Just reading it and deleting it. Not even opening it, just reading the title and thinking 'Oh that, here it comes again.' And that's why I'm saying, if Camosun had their themed up section, put them on a bulletin. I don't need to see them on my plate every day, because I'm not doing anything with them. They're useless. (Subject I, personal communication, November 1, 2007)

A few faculty members felt that e-mail was responsible for creating or furthering departmental conflicts. Faculty members interviewed from two different departments had very strong views about the inability of electronic communication to replace department meetings.

My department almost fell apart last semester because of an e-mail exchange that lasted two to three months. And by the time we had the face-to-face meeting, there was so much bad feeling, as a result of improper use of tone on e-mail or information that was merely rumour that was exchanged by e-mail and not corrected in person. And I think if the leadership in the department or in the school had actually stepped in and talked to people face-to-face, we wouldn't

have had that issue.... If the tone can be so misunderstood, and can come across as so much more forceful by e-mail than it would face-to-face, and I think that's what was happening. (Subject H, personal communication, October 24, 2007)

Another faculty member has had trouble using both Facebook and e-mail to communicate with colleagues. Because of the difficulty of finding a common time to meet, the department is attempting to discuss issues on-line through Facebook and is using e-mail for establishing meeting agendas. This person finds the lack of non-verbal signals and the medium itself allows for negative behaviour to occur.

My experience here on faculty is that there is an underlying theme of horizontal violence and bullying and it comes out in e-mails loud and clear, because the passive-aggressive stuff goes on there, so then it creates a rift and then no one wants to meet face-to-face because they don't want to conflict.

Then they go on e-mail and continue this kind of banter.

And it's sort of that detached being of the electronic safeguard that creates that moment, right? In a face-to-face you have all sorts of opportunities to dialogue back and forth and then to look at non-verbal (communication). But that's where we're moving and I'm finding it really troublesome that it's creating all sorts of rifts in the faculty and it's enhancing this culture that we've got in terms of passive-aggressive and bullying. It comes out in spades on e-mail. (Subject K, personal communication, October 23, 2007)

When I asked if there were any tasks for which electronic communication would not be used, three individuals responded they would not use the internet for departmental-wide discussions.

Electronic communication was seen as slower than other modes such as face-to-face or telephone, particularly when people take a long time to respond.

Right now, I particularly have experienced it at (name)
College, whereby my colleagues there, who aren't
particularly committed to the project we're doing although
their dean is, they don't reply. Ever. The only way I know
what they're thinking and what's going on is if I call the dean
or write to the dean or when we have our face-to-face
meetings, and sometimes they don't even come to the
table...And sometimes around here I wait a long time for
communication around here. (Subject N, personal
communication, October 31, 2007)

The time lag between sending a message and when it is read was seen as another way that speed was hampered by electronic communication. Issues that might have been addressed very quickly in a face-to-face meeting can be drawn out and take much more time to resolve.

Well it depends on when people read e-mails, right? And then how they're responding. We don't know right away. So we put it out there and somebody has not responded, like the 'stalkers' - they just go on, they read, they never post - and then they'll say "Well, that was really upsetting". So that was three weeks ago and you have a conversation now that that was an upsetting e-mail. Why wouldn't you have said

"(name), can we get together for a meeting?" or "Call me". (Subject K, personal communication, October 23, 2007)

It takes a fair amount of time to respond to e-mail. The second thing is it may actually take a lot more time to do something by e-mail than it would be to meet with someone and talk to them about it. Both in terms of the total minutes but also in terms of the duration that the process takes. (Subject M, personal communication, October 26, 2007)

A number of faculty members lamented the loss of face-to-face communication as more people chose to use electronic communication as their preferred medium. One person cited this factor as the reason why they felt that electronic communication technology did not increase their productivity overall.

Yes, significantly as I say, because every time you receive an e-mail communication ... I would say the majority, maybe not every time ... that's reducing your face-to-face contact. And education, as you know, is about inspiration. Most people are more inspired by face-to-face contact than they would be by a note from their instructor. (Subject H, personal communication, October 24, 2007)

Others appreciated the usefulness of the medium, but recognized that the time-saving aspect had a hidden cost in that they were reducing face-to-face contact with others.

You can deal with issues that you might have to meet with somebody about but you can deal with them much more quickly. You don't have to meet face-to-face. It does take away a lot of the face-to-face stuff but, if you want to be

productive, you just lose that, I'm afraid. You do lose it unfortunately. (Subject A, personal communication, October 29, 2007)

One individual felt that most electronic communication is taking place over e-mail and that productivity could be improved by switching to richer modes of communication.

The calendaring systems, scheduling systems, the websites, the blogs, the wikies, whatever we're trying to talk about, are a lot more richer and allow people to communicate, to coordinate things amongst themselves. I think this one-to-one e-mail, or one-to-many kind of e-mail thing is slowly becoming a thing of the past. (Subject I, personal communication, November 1, 2007)

As well, another person felt their productivity was reduced by an unreliable technical system at the college.

With the Camosun servers not sending through some of the good messages, that tends to hurt things because if a student sends a message and they don't get a reply, it's wasted two to three days before I find out, and who knows what can happen in that time? (Subject D, personal communication, October 22, 2007)

Interestingly, asynchronous communication was advanced as a factor that potentially decreased productivity. One person was involved in a project with faculty from a distant university and, while electronic communication was useful for transfer of information, it was frustrating at the same time.

It meant that you've got a team of six people, a group of six people, and because they're not responsible for the outcome in the sense they're not writing it, they're very quick to say "Last night at midnight, I had another idea." So, although I've already finished with that section, now we're back to other ideas. Things that wouldn't have happened in a meeting or, if they happened in a meeting, they'd have been noted and dealt with. So it just means there are lots of emails flying back and forth. It just means that I'm doing things at the 'eleventh hour', and that's not my style. It's easy for them to flip off yet another idea, but I'm the 'eleventh hour gal' and I don't like that. In that way, I don't like using e-mail. I'd rather have face-to-face. Now I can clarify through e-mail later, but I don't want more ideas that just pour in because someone got another idea. (Subject N, personal communication, October 31, 2007)

Other individuals had similar comments regarding how the asynchronous nature of electronic communication allowed individuals to pass tasks on to them without any agreement on their part.

I feel sometimes a bit impatient with students who ask me to take care of them. "Could you tell me all of my test results?" when that's their job to look after that themselves. So that can just set me back an hour worth of figuring stuff out, going back and finding it, blah-blah-blah, when I shouldn't have to do that. (Subject N, personal communication, October 31, 2007)

The very same thing I'm guilty of, others are, and sometimes it's just a big screen of people dumping stuff, and so they consider it done and yet it sits in your inbox as not done. So

I feel that pressure. (Subject J, personal communication, October 25, 2007)

One person described a situation where they received a number of emails from their departmental Chair asking for input into establishing criteria for student awards. Since information was missing from the original e-mail, it generated more e-mails:

So it just made a lot of work that maybe, in this case, a meeting would have sufficed. Or someone should just make a decision. Just a lot of extra communication went down that path. Just do your work, people. (Subject P, personal communication, October 30, 2007)

## **Electronic communication and overall Productivity**

When asked whether they felt that electronic communication made them more or less productive, eleven of the respondents said that electronic communication devices improved their productivity overall (Table 17). The asynchronous nature and its effectiveness for handling tasks allows one respondent to "send something that I need to have done and I can do that when I'm thinking of it, rather than have to try and carry it around in my memory and write it down and go deal with it the next day....And I believe that helps me stay more productive" (Subject J, personal communication, October 25, 2007) Other benefits of this technology noted were "sharing resources by e-mail" (Subject E, personal communication, October 29, 207), the ease with which students could "set up an appointment with me" (Subject F, personal communication, October 31, 2007) and the fact that "we have less department meetings because the

information just gets passed on in e-mails" (Subject O, personal communication, November 1, 2007).

Table 17. Overall, does ECT make you more or less productive?

	All faculty (16)	SOB faculty (9)	Other faculty (7)
More productive	11	7	4
Less productive	4	2	2
Not sure	1	0	1

*Note.* ECT = electronic communication technology. SOB = School of Business.

While one person felt that they were more productive overall, they were concerned that this productivity boost was at the expense of less face-to-face contact. This person knew some of their students from e-mail conversations and not from direct contact.

I hate e-mail. I hate e-mail. I hate Blackberries. I would not have a Blackberry if you bought me one. I think computers are great if you put them in their place, but I think they have really taken away ... they have increased efficiency, but they've taken away a lot of the face-to-face stuff. And I say that, not just for schools, but for kids socializing, all that kind of stuff. I think it's really had a devastating impact on the social lives of young people and how they relate to each other. It's funny, I go into my classroom and for the first three weeks, I made them put their nametags, name things in front, so I could figure out who they were, and I've got pretty big classes..... And so I will have a student come in and ask me questions and then, if she says "My name is Andrea," I'll say "Oh, Andrea Smith" and I'll know exactly

who she is, because of how many times she's e-mailed. I don't know her in the classroom. I'd say I have about ten of those students. When they say "My name is Christopher Whatever," I say "Oh, Christopher Blah-Blah-Blah? Okay, I know exactly everything about you. I know your personal life, I know all the problems you're having", I know everything from the e-mails. But I've never met this kid. (Subject A, personal communication, October 29, 2007)

Four participants felt that they were less productive due to the use of electronic communication technology. One person said that electronic communication "snuffs out my creativity" (Subject K, personal communication, October 23, 2007) and they had defined productivity as the ability to be creative. A similar view was shared by another faculty member who feels that most e-mail is "reducing your face-to-face contact. And education, as you know, is about inspiration. Most people are more inspired by face-to-face contact than they would be by a note from their instructor" (Subject H, personal communication, October 24, 2007). Another said that while there is some benefit due to its asynchronous nature, the volume of e-mail makes it unproductive.

There is the timing thing. If you're here in the morning and I'm here in the afternoon, there is some benefit to that. But I bet, for every e-mail that's like that, there are five that aren't. And so I think that it's just too many e-mails for that small productivity boost. (Subject P, personal communication, October 30, 2007)

That sentiment was shared by the fourth participant who said "spending less time on e-mail has definitely increased my productivity and improved my quality of life" (Subject L, personal communication, October 25, 2007).

One respondent was unsure of the impact of technology on their productivity as they are in the 21 to 30 age range and have never worked without technology. "So I have sort of always been a part of that since I'd left high school. So I don't really know what it would be like without it" (Subject G, personal communication, October 23, 2007).

#### Personal strategies for improving Productivity

Participants discussed many different techniques they have found helpful in improving their productive use of electronic communication. The most popular strategies discussed were use of folders for storing e-mails, and limiting the number of times they checked for new messages (Table 18). "You know how you have it, most people have it so that, if an e-mail comes in, it flashes that an e-mail's in, I have that turned off. I think that has helped my productivity, when I changed that" (Subject P, personal communication, October 30, 2007).

Deleting a lot of the e-mails, often without reading the body of the message, was also seen as a popular choice. "I find it just takes up too much time. And I'm not prepared to do that. I'm not going to sit in my office hour reading e-mails that doesn't (sic) relate to my work, and I don't even save them for later" (Subject L, personal communication, October 25, 2007).

Table 18. Strategies employed which improve participants' productivity

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Use folders to store e-mails	7	4	3
Check only a few times each day	6	3	3
Delete a lot, often without reading	4	3	1
Turn off automatic e-mail	3	1	2
Encourage meetings, face-to-face	3	1	2
Use Outlook tools	3	2	1
Reply immediately	2	0	2
Check e-mail all the time	1	1	0

*Note.* More than one answer per person is possible. SOB = School of Business.

Other strategies suggested were to use Outlook tools, check e-mail frequently, reply immediately, and finally, to improve productivity by encouraging meeting instead of exchanging electronic communications.

I really hate it too actually, in terms of the office environment with my colleagues. I really dislike e-mail, unless it's just purely an informational thing, because it's so limiting. You can't really decide anything that's complex or complicated that way. It just takes forever, with hundreds of e-mails going back and forth before you're done, when you could have just walked twenty feet and had a conversation and solved the whole thing. (Subject M, personal communication, October 26, 2007)

#### College-wide strategies for improved electronic communication usage

Most of the persons interviewed had suggestions to make regarding improved usage of electronic communication technology at Camosun College (Table 19). Many of the ideas involved reducing the volume of e-mail, such as

less 'cc'ing' of e-mails, use of bulletin boards, and better spam filters. "Obviously, continuing to reduce or eliminate spam, so we don't have that many more things to click on, when in fact it's such a major part of our productivity, that would be useful" (Subject J, personal communication, October 25, 2007).

Table 19. Strategies suggested for improving ECT at the College

	All faculty	SOB	Other
	(16)	faculty(9)	faculty (7)
More limited use of 'cc' feature	4	2	2
Use bulletin boards more and e-mail	3	1	2
less			
Better spam filters	3	2	1
Departments able to adopt own IT	2	2	0
systems			
Make student e-mail addresses	1	0	1
available to faculty and staff			
Increased storage space for e-mail	1	0	1
Centralized IT access for students	1	1	0
Common system for both campuses	1	1	0
Be informed when an e-mail is	1	0	1
forwarded			
Importance of e-mail is obvious from	1	1	0
subject line			

*Note*. More than one response per person is possible. ECT = electronic communication technology. SOB = School of Business. IT = information technology.

Other suggestions were to get removed from group e-mail lists, be informed when one of their e-mails was forwarded, and to be able to ascertain the importance of the subject by reading the subject line. "But do I need to be on a list for every instructor in the college? I don't think so" (Subject P, personal communication, October 30, 2007). "Don't just 'cc' to a whole bunch of people. I don't know if this is appropriate or not, but sometimes I have no interest to know anything in other departments" (Subject F, personal communication, October 31,

2007). Three people suggested bulletin boards for dissemination of college-wide or departmentally relevant messages.

If I wanted to see something about the College community, I can go to see it, and I don't need to keep deleting these messages, because I hardly ever look at that....And that's why I'm saying, if Camosun had their themed-up section, put them on a bulletin. I don't need to see them on my plate every day, because I'm not doing anything with them.

They're useless. (Subject K, personal communication, October 23, 2007)

#### Time logs

The data collected in the time logs has been assembled into a number of tables and is presented throughout the following section. Additional summaries of e-mails received during the work week and the handling of old e-mails are presented in Tables F5 and F6 respectively in Appendix F.

#### Volume of e-mail

Analysis of the time log data suggests that faculty members are sending few e-mails each weekday, with the number of work-related electronic communications sent ranging from 0 to 8, with just over 1 message sent on average by all faculty members (Table 20). While the School of Business faculty send almost twice as many electronic communications as other faculty on weekdays, one individual within School of Business sent one-third of the e-mails for the entire faculty group. Very few e-mails were generated by any faculty members on the weekends.

Table 20. Number of electronic communications sent by participant

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Range	0 to 8	0 to 8	0 to 4
Average	1.3	1.6	1

Note. Table indicates the range and average number of e-mails sent on an average week day by individuals in each grouping, with averages rounded to the nearest tenth. SOB = School of Business.

A comparison of the average number of electronic communications received each workday showed a difference between the School of Business faculty and other faculty, with School of Business faculty receiving 9 communications and other faculty receiving 15 communications on an average workday (Table 21). This discrepancy could partially explain why only 40% of communications received by other faculty members was considered to be 'very important' to their work when School of Business faculty felt that 48% of similar communications were 'very important' to their work. If the communications that participants had rated as 'somewhat relevant' are included with the 'very important' messages, then these messages account for 68% of e-mail volume received by other faculty and 70% for School of Business faculty.

These relatively low rates may explain why a number of faculty members cited e-mail as a waste of time. Roberson's (2004) study, which collected time logs for individuals from a wide-range of organizations, found that 28% of e-mails received were considered 'very important' to work, with that number increasing to 60% when 'somewhat important' e-mails were included (p.75). While only seven of the 45 individuals interviewed by Roberson (2004) completed the time logs, the findings are similar to those generated by this study (p.76).

Table 21. Number of work-related e-mails received during the week

	Very relevant to my work	Somewhat relevant to my work	Unsure of relevance	Not relevant to my work	Total
SOB faculty (9)	194	90	39	82	405
Daily average per person					9
Other faculty (7):	209	149	59	107	524
Daily average per person					15

*Note.* Number of work-related e-mails received by all participants from Monday to Friday. SOB = School of Business.

#### Average time spent each day on electronic communications

The average amount of time spent each day on electronic communication by all participants is 27 minutes from the office and an additional 10 minutes from a remote location (Table 22). School of Business faculty's average time from a remote location is considerably higher than other faculty's time, with 14 minutes on average compared to 4 minutes. Weekend behaviour is similar between the two groups, with few e-mails being initiated or read, but again with more time spent by School of Business faculty than other faculty.

Table 22. Average time spent on work related electronic communications

	All faculty (16)	SOB faculty(9)	Other faculty (7)
Weekday average from work site	27 minutes	27 minutes	26 minutes
Weekday average from off- site	10 minutes	14 minutes	4 minutes
Weekday total average	37 minutes	41 minutes	30 minutes
Weekday total range	0 to 90 minutes	0 to 90 minutes	5 to 90 minutes
Weekend average	10 minutes	13 minutes	5 minutes
Weekend range	0 to 60 minutes	0 to 60 minutes	0 to 30 minutes

*Note.* Averages represent the average amount of time spent per individual within each group, rounded to the nearest minute. SOB = School of Business.

### Categorization of e-mails by relevance to work

Participants were asked to categorize e-mails received in terms of relevance to their work and to identify actions taken for each e-mail. The time log data indicates that faculty members spent time on e-mails that they had rated as not relevant to their work – they deleted the item after reading it, replied, forwarded it, or took some other action (Table 23). Of all electronic communications received, 11% of the School of Business faculty's and 6% of the other faculty's e-mails fell into this category, suggesting that time is spent on communications that bring little or no value to the recipients.

While stopping this behaviour looks like an obvious and easy way to improve productivity, it may not always be possible to determine the relevancy of a communication without first reading it. Additionally, while this study considers the relative volume of e-mails, it does not identify the percentage or amount of time spent on irrelevant communications.

Table 23. Percentage of irrelevant e-mails on which time is spent

	SOB faculty (9) E-mails not relevant to my work	Other faculty (7) E-mails not relevant to my work
Deleted after reading	22	18
Replied and/or forwarded	20	. 13
Other action taken (phoned, met with person, etc.)	1	
Total of above e-mails	43	31
Total e-mails received	405	524
Irrelevant e-mails on which time is spent as a percentage of total e-mails received	11%	6%

*Note.* Other than 'percentage of total', amounts represent the number of e-mails received during the 5-day work week participants graded as 'not relevant to their work' as categorized by the action taken. SOB = School of Business.

Data was gathered on the number of electronic communications received during the 5 week days of the study that participants rated as 'very relevant', 'somewhat relevant' or they were 'unsure of the relevance' to their work. Both groups deleted without reading or took no action on 12% to 13% of items received that week that they said were 'very relevant', 'somewhat relevant', or 'might be relevant' to their work (Table 24).

Table 24. Summary of e-mails deleted with no action taken

	1 - 2			
	Very	Somewhat	Unsure of	Total
	relevant to	relevant to	relevance	
	my work	my work	to my work	
School of Business faculty (9):				
Deleted without reading entire	2	4	27	33
body of message				
Took no action (didn't read body	7	9	2	18
of message, didn't delete)				
E-mails with no action taken	9	13	29	51
E-mails on which action taken <sup>a</sup>	<u>185</u>	77	<u>10</u>	<u>272</u>
Sub-total	194	77 90	39	323
E-mails which were not relevant <sup>b</sup>				<u>82</u>
Total e-mails received				405
Relevant or potentially relevant e-mails on which no action is taken as a				
percentage of total e-mails				13%
Other faculty (7):				
Deleted without reading entire	0	18	25	43
body of message				
Took no action (didn't read body	2	13	7	22
of message, didn't delete)				
E-mails with no action taken	2	31	32	65
E-mails on which action taken <sup>a</sup>	207	<u>118</u>	<u>27</u>	<u>352</u>
Sub-total	209	149	59	417
E-mails which were not relevant <sup>b</sup>				<u>107</u>
Total e-mails received				524
Relevant or potentially relevant e-mails on which no action is taken as a				12%
percentage of total e-mails				

Note. Other than percentages, amounts represent the number of e-mails received during the 5-day work week participants graded as 'very relevant', 'somewhat relevant', or 'unsure of relevance' that they deleted or took no action on.

<sup>b</sup>Total of all e-mails that were included in the 'not relevant to my work' column.

When communications for which participants were unsure of their relevance are removed, no action was being taken on 5 or 6% of the e-mails that were graded as relevant to their work (Table 25). Roberson's (2004) study of e-mail within organizations found 8% of e-mails judged to be very or somewhat important to work were deleted without reading or were left unread. This finding

<sup>&</sup>lt;sup>a</sup> Total of all other rows from time log where action is taken on e-mails received that were graded as 'very relevant', 'somewhat relevant', or 'unsure of relevance'.

suggests that this form of communication may not be entirely successful for the dissemination of important information.

Table 25. Relevant e-mails deleted with no action taken

	Very relevant my work		Somewhat relevant to my work	Total
School of Business faculty (9):				
Deleted without reading entire body of message	2		4	6
Took no action (didn't read body of message, didn't delete)	7		9	16
E-mails with no action taken		9	13	22
E-mails on which action taken <sup>a</sup>		<u> 185</u>	<u>77</u>	<u>262</u>
Sub-total		194	90	284
E-mails from other categories <sup>b</sup>				121
Total e-mails received				405
Relevant e-mails on which no action is taken as a				5%
percentage of total e-mails				L
Other faculty (7):	T -			
Deleted without reading entire body of message	0		18	18
Took no action (didn't read body of message, didn't delete)	2		13	15
E-mails with no action taken		2	31	33
E-mails on which action taken <sup>a</sup>		207	<u>118</u>	<u>325</u>
Sub-total		209	149	358
E-mails from other categories <sup>b</sup>				<u>166</u>
Total e-mails received	<u> </u>			524
Relevant e-mails on which no action is taken as a percentage of total e-mails			6%	

*Note.* Each category indicates the total number of work-related e-mails received by all participants in each grouping during a 5-day work week.

A review of the logs and the comments made during the interviews suggests that most instructors deal with electronic communication shortly after receipt or not at all. Dealing with old electronic communications, defined as ones received prior to the day, varied. Some individuals spend no time at all on old e-

<sup>&</sup>lt;sup>a</sup> Total of all other rows from time log where action is taken on e-mails received that were graded as 'very relevant' and 'somewhat relevant'. <sup>b</sup>Total of all e-mails that were included in the 'unsure of relevance' and 'not relevant to my work' columns.

mails, preferring to deal with all incoming mail each day. One person spends about one hour each week cleaning out old e-mails, while others were only dealing with old e-mails on Monday, suggesting that the generation of these items had been over the weekend. One person dealt with old e-mails every workday and another dealt with old e-mails every day. This variety suggests that while it may be difficult to predict when a response is forthcoming, e-mails sent will likely get an immediate response or no response.

It would have been logical to assume that there would be a correlation between behaviour in terms of dealing with old e-mails and the number of e-mails in the inbox. Unfortunately, that was not the case, as some people who deal with all their e-mails as soon as they are received leave everything in their inbox during the term. Most of the faculty members interviewed had less than 30 e-mails in their inbox on the day that they were interviewed. The remaining individuals had over 100 e-mails saved in their inbox, with three faculty members having more than 500 e-mails in their inbox. One person whose e-mail volume fell in the 100 to 400 category said that increased workload led to this atypical volume and less than 30 e-mails was their norm.

#### Changes to behaviour patterns

Four of the participants stated that they changed their pattern of dealing with electronic communication when completing the time log portion of my study. Interestingly, all four were women, representing half of the women interviewed.

Of the four women who changed their behaviour, three commented that they made a point of spending less time on e-mail than they would have normally, with

one woman consciously spending less time checking her e-mail on the weekend.

One person felt it would be easier to count the number of times she checked her e-mail by closing her e-mail account, although normally she would have had her e-mail account open all day.

Two women commented that they used the time log to reinforce a change in their e-mail behaviour that they had been planning to make for a while. One woman indicated that she had changed her behaviour "a little bit, but not more than I had intended to. As I said before, I've been - there's been sort of a program of scaling back use" (Subject G, personal communication, October 23, 2007). One woman said because of her involvement in my study, she was more thoughtful about sending electronic communications and that she now considered communicating face-to-face before responding electronically.

The time log helped me to realize which ones I had to respond to and which ones I could actually just file or talk to the person, so it helped me to be conscious of the communication that was coming in, which was great.

(Subject H, personal communication, October 24, 2007)

One woman changed her pattern by deleting e-mails after she had dealt with them. These changes support Giddens' (1984) contention that communication may be altered by the participant, by the modality, or by both due to their recursive relationship.

Another gender issue relates to the volume of e-mails received. The women in the study received almost 40% more e-mail during the work week than their male counterparts, with little difference noted on the weekends (Table 26).

It is unclear what is driving this additional volume; there was no difference noted in terms of the number of e-mails sent by women as compared to men studied. This study did not determine from whom e-mails were received and so it is unclear whether this volume is generated by students, administrators or other faculty members.

Table 26. Comparison of the volume of e-mails received by gender

	Female participants (8)	Male participants (8)
Work-related communications	Total = 545	Total = 389
received today - Monday to Friday	Weekly Average = 68	Weekly Average = 49
Work-related communications	Total = 41	Total = 43
received today - Saturday and	Weekly Average = 5	Weekly Average = 5
Sunday		

*Note.* Averages were calculated by dividing each total by the number of participants in each group.

# **CHAPTER 5: LITERATURE REVIEW REVISITED**

The literature reviewed in Chapter 2 examines faculty use of electronic communication at primarily English-speaking colleges and universities, in terms of communication, effectiveness, gender, and time issues. As a method of testing the validity and reliability of the study results, the study findings are compared with these corresponding studies, and any similarities and differences reported. In addition to those findings that correspond with the literature reviewed, other emergent results are discussed, and any consistencies between the study results and Giddens' (1984) Structuration Theory are noted.

#### **Communication Issues**

The literature on electronic communication highlights issues regarding this medium and its functionality at educational institutions. One issue is clarity of communication; a number of studies suggest that electronic communication creates confusion and that other communication devices, such as telephones or face-to-face conversation, are better suited for complex topics and equivocal tasks (Atamian & DeMoville, 1998; Duran et al., 2005; Haworth, 1999; Ishii, 2005; Montano & Dillon, 2005). Clarity of communication was definitely mentioned as a concern for participants in my study. Electronic discussions had exacerbated conflicts and resulted in hurt feelings. As a lean medium, "the tone can be so misunderstood, and can come across as so much more forceful by e-mail than it would face-to-face" (Subject N, personal communication, October 31, 2007).

As well, clarity of communication was seen as an issue with foreign students who are non-native English speakers as "sometimes their English is not very good....And sometimes it's a little hard to understand what they're asking, because they have difficulty composing their message" (Subject B, personal communication, October 22, 2007). One half of participants felt that electronic communications from English as a Second Language students were different, with the main difference being difficulty in understanding their requests. Chen's (2001) study involved a comparison of the actual wording of the e-mails sent by students to their professors. This study did not look at actual e-mails between faculty and their students. Therefore, while the differences that Chen found may exist, they were not revealed in this investigation.

The question regarding dissimilarities in electronic communication with older students did elicit answers that support Chen's (2001) study. While the interview questions did not specifically ask about writing style, some answers about older students obliquely referred to this topic. Different communication styles were exhibited in the e-mails sent by older students as compared to younger students.

Tapscott (as cited in Pletka, 2007) coined the phrase 'Net Generation' to characterize the group born between 1982 and 2002. Net Generation students are unlike earlier generations; "they are digital natives in a world of evolving information and communication technologies made for their interaction with all sectors of society" (Pletka, 2007, p.20). Having grown up with computers, they are used to the immediacy of information; where older generations perform

research at the library, the Net Generation has been trained to surf the web (Tapscott, 1998). A lot of their communication is text-based, including text messaging on cell phones, engaging in web-based conversations, and e-mail.

This phenomenon may be responsible for the style changes being observed, with some faculty members commenting that communications from older students tend to be "more formal than what some younger students would write. They might be less likely to say 'Hey' or use all lower case letters or that sort of thing" (Subject G, personal communication, October 23, 2007). Duran et al.'s (2005) study also mentions that some professors feel students' on-line behaviour is inappropriately casual.

These communication-style changes are consistent with Giddens' (1984) duality of structure notion, suggesting that agents are influenced by the structure of the organization within which they are operating and may alter the structure through their behaviour. Giddens' (1993) proposes that individual acts may cause social change but that social structure will only change when the new methods are institutionalized and thus form part of the social structure, suggesting that policies regarding electronic communication need to be formalized in order for an organization, such as Camosun College, to influence the changes that are occurring.

Participants reported that student communications were related primarily to course content and issues regarding missed classes. "So either actual content of the class or questions to do with the structure – 'When will this take place?' or 'What will be covered' on a certain test or things like that" (Subject D, personal

communication, October 22, 2007). The content of e-mails received was not studied and so this assertion is based upon respondents' answers to interview questions rather than an examination of the communications. This finding appears to be consistent with communications at the small, private institution in Duran et al.'s study (2005), as opposed to communications at the larger, public university, where students enquired mainly about grades.

Duran et al. also found that certain topics may not be well suited to electronic dissemination; this study supports this finding. Fifteen participants stated that they prefer to employ either face-to-face or telephone communication for sensitive or personal issues. "I would not use e-mail if I had to say something very private to someone, or something that had some emotional attachment to it, like I was upset with someone or if it was daunting" (Subject A, personal communication, October 29, 2007). Only one individual felt that electronic communication could be used for any discussion, including arguments or sensitive matters.

There are some cases where I've seen someone face-to-face rather than writing an e-mail, but that's not necessarily because I'm against writing an e-mail, it's just because they're two doors down from me and it's easier to get an answer right away. I don't think there are very many things that I would outright refuse to use e-mail. (Subject D, personal communication, October 22, 2007)

#### Time Issues

No participants said that electronic communication technology increased their workload, despite frequent comments such as "it takes a fair amount of time to respond to e-mail" (Subject M, personal communication, October 26, 2007). This finding is consistent with Haworth's (1999) suggestion that e-mail did not increase the volume of student-faculty interaction; instead, it redistributed contact to an alternate form.

The findings are also consistent with Duran et al's (2005) study, which suggests the extent to which e-mail is augmenting or replacing other forms of communication is dependent upon instructor and student variables, such as personality and communication preferences (p.173). Certainly the amount to which e-mail is being used to replace or support face-to-face conversations or telephone calls varies by instructor and is determined to a large part by individual preferences.

A few studies suggest that office hours may no longer be adequate to meet students' needs and that electronic communication may be a replacement (Atamian & DeMoville, 1998; Haworth, 1999). Eleven of the interviewees felt students' use of their office hours had decreased due to electronic communication. "I think it reduces it, because they'll e-mail me in preference to stopping by my office. It's more immediate, so they can generally get answers more quickly than having to wait for an office hour" (Subject B, personal communication, October 22, 2007). The results of this study corroborate the findings of Atamian and DeMoville (1998) and Haworth (1999) on this point.

It is surprising to note that none of the participants complained about the amount of time they are spending on electronic communication in the evenings, on weekends, and even on holidays. Many academics appear to enjoy the flexibility offered by academia and adopt non-traditional working hours more conducive to their creativity. Electronic technology facilitates and enhances flexible work schedules and provides the opportunity to work off-campus. It is worth noting that all of the volunteers willingly agreed to an interview that could take up to an hour and a half, to devote time every day for seven days to complete a time log, and to review a transcript of their interview with no compensation. It was outside the bounds of the study to determine whether other instructors at the College would share this phenomenon.

#### Gender and Age Issues

Duran et al.'s (2005) study identified a gender difference in terms of electronic communication, with female faculty receiving more student e-mail than male faculty. Due to the qualitative nature of this study and the small number of individuals interviewed, a decisive supportive claim which would correlate Duran et al.'s study cannot be made. While this study, unlike Duran et al.'s work, did not distinguish between student e-mails and all other e-mails received, the eight female participants did receive 40% more e-mail during the work week than the eight male participants. Both groups received similar numbers of e-mails on the weekend and both groups sent similar numbers of e-mails. While the size of the sample base for this study did not allow for a deeper exploration of this issue, it does appear to be an area rich for further inquiry.

Another gender difference identified in the study is that half of the women, and none of the men, adjusted their behaviour when completing the time log. "I discovered a few things....I consciously spent less time checking e-mail on the weekends" (Subject J, personal communication, October 25, 2007). As well, of the four individuals who felt less productive as a result of electronic communication, three were women. These findings differ from Ogan and Chung's (2003) study which determined that gender did not affect faculty use of computers and computer-based technologies. Again, the small sample size may be an issue here in terms of making a definitive conclusion, though the finding does support Giddens' (1984) notion of the interaction between structure and agent and the ability of both to change recursively.

Certain studies look for a connection between age of faculty members and their ability to utilize electronic communication technology (Ogan & Chung, 2003; Wang & Cohen, 1998). Again, given the qualitative nature of the study and relatively few participants interviewed, strong assertions related to age are problematic, but age did not appear to play a major role. One might postulate that the age of the instructors would impact the perceived benefit of the technology, but this assertion cannot be supported by these findings. While very comfortable with technology, the two youngest participants felt that certain aspects of it diminished their productivity to some extent. Of the four instructors who said productivity was reduced overall by technology, two are in the 31 to 40 age range and the other two are between 41 and 50 years of age. All of the participants in the oldest category felt that their productivity was enhanced by

electronic communication technology. Wang and Cohen (1998) could not determine whether use of electronic communication varied with age and this study also suggests age is not a major factor in use of communication technology.

#### **Productivity**

Various studies propose that faculty productivity includes both quantitative and qualitative factors and is composed of such activities as advising, teaching, research and scholarly activity, and community involvement (Bock, 1997; Katula & Doody, 1990, Middaugh, 2001). Participants in the study cite similar components in their personal definitions of productivity. Creating new course material, aiding student progress, helping on college or departmental projects, personal learning and growth were some of the main areas mentioned. These definitions seem more in line with traditional notions of performance versus the more concrete, quantitative definitions of productivity.

Most instructors felt electronic communication enhanced their overall productivity, citing benefits such as its use for quick, procedural tasks and for asynchronous communication. "Because I can share things with so many people so easily. I can respond to things quickly and when I feel it's useful to do it, or in the right time to do it" (Subject E, personal communication, October 29, 2007).

They also remarked, however, on aspects of electronic communication that hinder their ability to do their job.

Well, there's a trade-off between e-mails that are meaningful and help you to be productive and the ones that just clutter your inbox and that you have to wade through all the time....For me personally, it's a positive trade-off. On balance, it's a good thing. (Subject B, personal communication, October 22, 2007)

Some other instructors feel that their overall productivity is diminished by electronic communication technology.

Well, if you take productivity as your effectiveness as an instructor, then there was a period of time when I found it really distracting and it got in the way of teaching....I think it gets in the way of the face-to-face contact. (Subject H, personal communication, October 24, 2007)

Many of the studies recommend faculty development and staff training for more effective use of technology (Amey & VanDerLinden, 2003; Brown et al., 2004; Shepherd, 2004; Wang & Cohen, 1998). The need for technical training was mentioned in the interviews. "I think it has the potential to make me far more productive. If I was more familiar with it. And what I struggle with is finding the time to be more productive with it" (Subject C, personal communication, November 2, 2007). Other comments made by participants suggest that some of the problems they are encountering with electronic communication could be resolved with appropriate staff training.

We spend hours with students the first few weeks, setting up principles of how we're going to respect each other, trust each other, what happens when people challenge the status quo, how we're going to react, conflict guidelines in place to launch groups so that we can make learning happen in the

classroom. We don't do that as faculty face-to-face, nor do we do that on e-mail. (Subject K, personal communication, October 23, 2007)

One difficulty with electronic communication is that different approaches and comfort levels with the technology exist within the organization (Roberson, 2004). Part of the potential for conflict or lack of increase in productivity is the use of a modality that is viewed differently by different players. Major differences exist in terms of response and access times and in terms of suitability of the medium to all tasks. Some faculty members happily embrace technology and use it for everything while others do not see it as useful for some tasks and consider it detrimental for other jobs. One person within the study, for example, did not want to discuss departmental issues over the internet and yet that person's department had decided to use both Facebook and e-mail as replacements for meetings.

#### Conclusion

The findings of this study support most of the findings of the literature reviewed. In particular, this study also found that clarity of electronic communication can be problematic, communication with older students tends to be different, and that this modality may not be best suited for certain topics, such as contentious or complex issues (Atamian & DeMoville, 1998; Chen, 2001; Duran et al., 2005; Haworth, 1999; Ishii, 2005; Montano & Dillon, 2005). In terms of time, most participants felt that use of communication technology re-distributed rather than increased their workload, with many experiencing a reduction in

student use of office hours (Atamian & DeMoville, 1998; Haworth, 1999). This study also found a gender difference in terms of volume of e-mails received and behavioural changes of female participants, consistent with Duran et al.'s (2005) study and in contrast to Ogan and Chung's (2003) results. Age was not seen as a major factor in determining use of technology among faculty (Wang & Cohen, 1998). Giddens' (1984) Structuration Theory is supported by the noted shift in writing style adopted by some students and by the change in behaviour of some faculty in terms of their use of the technology.

The over-arching question addressed in this study is: In the perception of college faculty, what impact does electronic communication technology have on their performance? The majority of participants felt that electronic communication enhanced their ability to do their work and that, overall, the technology improved their performance. Electronic communication was cited as useful for procedural tasks, for asynchronous and quick communication, and for encouraging contact with shy or reticent students. Participants had many suggestions for improving use of technology at the College that could further enhance their productivity and some of those ideas, along with ones based upon the findings of this study, will be discussed in the following chapter.

# CHAPTER 6: RECOMMENDATIONS & SUGGESTIONS FOR FURTHER RESEARCH

Colleges and universities are operating within what has been called the knowledge economy; whereas natural resources and physical inputs were essential in the preceding post-industrial era, information and intellectual skill now take precedence (Powell & Snellman, 2004). One aspect of this shift is the enormous amounts of information now readily available to educators. The handling of such information is an essential element of daily tasks, and how well faculty deal with communication media such as e-mail, website postings, and text messaging has a significant impact on their productivity. Giddens (1991) posits that new information requires reflection or "chronic revision in the light of new information" (p.20), which suggests that educational institutions should be adaptive and thoughtful about their processes regarding information. Colleges, out of necessity, tend not to look at improving their handling of information, but rather focus on structural changes that deal with economic and financial elements (Fisher & Rubenson, 1998). Such a focus may lead administrators to miss out on an important arena where gains in productivity could be easily realized. My recommendations involve issues of strategic changes, policy changes, and the need for staff training.

### Strategy

It is worth remembering that electronic communication is a relatively new medium. In the Australian study of the impact of technology on faculty performance, use of computer-based learning, such as websites and e-mail, increased from 5% in 1994 to 42% in 2004 (Krause et al., 2005). A number of individuals in my study indicated that their usage of electronic communication has evolved and changed over time, and will probably continue to evolve. While they previously used e-mail extensively, some participants now utilize it as little as possible.

It's been an evolution. I was one that was loving e-mail when it first came out. It was a new toy, I used it like crazy, got caught up in the craze about it and, yes, I probably was a contributor to some of those bad e-mails as well....And then reflecting on that has pushed me in a completely different direction to e-mail. (Subject K, personal communication, October 23, 2007)

This evolution may continue to the point where e-mail is no longer the main form of electronic communication. Another participant feels that e-mail is archaic:

The calendaring systems, scheduling systems, the websites, the blogs, the wikies, whatever we're trying to talk about, are a lot more richer and allow people to communicate, to coordinate things amongst themselves. I think this one-to-one e-mail, or one-to-many kind of e-mail thing is slowly becoming a thing of the past. (Subject I, personal communication, November 1, 2007)

Some people had already modified their usage of electronic communication. Others said they want the College to send fewer e-mails, using bulletin boards instead. A colleague mentioned that a notice to change departmental meeting times was posted in every weekly departmental e-mail for six months and yet there were still faculty members who did not know about the change. Certainly, the fact that participants deleted e-mails without reading them when they might be relevant to their work suggests this method may not be effective for the sharing of information. "If I'm feeling involved or want to get involved in college community, put it on a bulletin board, I'll go look at it" (Subject K, personal communication, October 23, 2007).

As instructors are clearly choosing to ignore e-mailed messages, it appears that this is not a good medium to use for dissemination of general information. Instead, a more appropriate forum might be via methods whereby the instructor chooses to access the information, such as bulletin boards. While this technique would reduce e-mail volume, and might increase the probability that those e-mails received will be read, it is unknown whether or not faculty members will actually read the general information posted to the bulletin boards.

Bulletin boards should be utilized for electronic communications disseminating general information intended for large groups within the college community.

#### Policy

Giddens' duality of structure notion suggests that agents may be making changes to the college structure through their behaviour (Macintosh, 1994). By

sending e-mails on the weekend, students are shifting faculty behaviour from one where faculty were only available during office hours to one where access is less restricted. Use of language is also changing, with communications from younger students being less formal in their writing style. It may be time to formalize these structural changes and give some control of the change back to the College and faculty.

The results of this study point to the need for additional policy regarding use of electronic communication at the college. The existing College policies do address general issues of computer usage, such as privacy of data, respect of other computer users' privacy, and appropriate use of computing resources (Camosun College, 2003a, 2003b) The Student Conduct Policy gives examples of unacceptable student conduct and discusses the ramifications for students who disobey the policy (Camosun College, 2007). The Student Conduct policy covers electronic communication indirectly, stating, for example, that "any complaint that is intentionally made by a person who knows it is false, frivolous, or vexatious" violates the policy (p.38). Obviously, electronic communications could fall into that category yet more explicit directions should be given regarding on-line conduct, including language and usage. Many faculty members mentioned inclusion on mailing lists as harmful to their productivity; the policy could suggest verbal or written permission be obtained prior to their name being added to a list. A search of a neighbouring university's website disclosed a document addressed to its students outlining appropriate on-line behaviour or

'netiquette'; Camosun College should consider posting a similar document to its website (Royal Roads, 2007).

In order to minimize confusion and inappropriate behaviour, colleges and universities should develop and disseminate clear policies regarding electronic communication usage, establishing norms and protocols that are understood by all users.

Within the study, there was a wide range of beliefs about what was an acceptable response time to student electronic communications, ranging from less than 6 hours to 3 days. In one interview, a situation was discussed where School administrators had implicitly agreed that it is reasonable to expect a faculty member to check their electronic communication messages within a certain number of hours. That decision, which was upsetting to the instructor, does not appear to have been based upon either a School-wide or College-wide policy.

Universities and colleges should establish policies regarding appropriate response times to student and other work-related electronic communications. While an institution-wide policy regarding all response times is recommended, instructors' right to decide their own, individual policies regarding response times for weekends and vacation should be considered.

Despite the varying response times to electronic communications amongst faculty members, there is very little communication to students or others regarding when they can expect a response. One faculty member found it

frustrating to not know when a response from a co-worker could be expected, saying that they didn't "want to wait five days for something I should have known yesterday. If we are going to use e-mail to communicate, then we need to do it" (Subject N, personal communication, October 31, 2007). And yet by not indicating when a response is likely, that is the situation that students are in if they send electronic communications to some faculty.

Instructors should be encouraged to state their particular policy regarding response times to electronic communications on their course outlines, including expected response time over weekends and holidays.

#### Training

Electronic communication has the capacity to enhance faculty productivity and performance but this capacity is not being fully realized. On average, participants spend the equivalent of almost one full workday a week on electronic communication. Even a small reduction in time spent on this activity could have a huge impact on the overall productivity of the institution. Participants remarked on areas in which they felt they could make improvements regarding their use of electronic communication. Some of the recommendations they made were that they limit the number of times they check their e-mail and stop reading communications that have little value.

I get newsletters that are marginally useful. I spend some time reading that and keeping up, but it's more, yes, I read those. They're interesting but they take some time. It's partly e-mail, partly web-surfing. Once the thing's up, you

know. (Subject E, personal communication, October 29, 2007)

The usefulness of technology at the College could be improved with training (Amey & VanDerLinden, 2003; Brown et al., 2004; Shepherd, 2004; Wang & Cohen, 1998). One person mentioned that their department invests a considerable amount of time training their students regarding interactions and behaviour but no staff training on electronic communication has taken place.

There are not really any clear principles about creating learning environments. We spend hours with students the first few weeks, setting up principles of how we're going to respect each other, trust each other, what happens when people challenge the status quo, how we're going to react, conflict guidelines in place to launch groups so that we can make learning happen in the classroom. We don't do that as faculty face-to-face, nor do we do that on e-mail. (Subject K, personal communication, October 23, 2007)

In order to enhance use of system capabilities and therefore productivity, training should focus on appropriate electronic communication usage, on-line behaviour and technical skills. Such education could, by increasing user sensitivity and awareness, help reduce potential conflicts. Training may also reduce some of the behaviours that were noted as annoying by some participants, such as unnecessary copying on e-mails and inclusion on mailing lists that are of no interest to them. A fair number of conflicts arose around the use of electronic communication devices for handling complex issues.

With (name)'s list of awards - it's saving us time maybe, because we haven't had to have a meeting, but there wasn't enough information on that for me to do any work with it.

There's no dollar amount. It wasn't clear to me, because I'm the 'newbie', what those eight lists was supposed to be matched up with, and so it just made a lot of work that maybe, in this case, a meeting would have sufficed. (Subject P, personal communication, October 30, 2007)

Ishii (2005) suggests that training may improve the efficacy of electronic communication's use for more complex or equivocal tasks. Part of that training may be learning to identify which tasks are better suited to richer media, such as face-to-face or verbal communication.

Maintaining the time log highlighted for several faculty members the amount of time spent on potentially unproductive or less productive communications and they have since adjusted their online behaviour. It is likely that similar benefits could be gained by other faculty members through introduction of a time log exercise. Faculty could be encouraged to track their time spent on electronic communications for a short period of time. This low-cost, relatively simple exercise could begin the process of better time management of electronic communications

The benefit of electronic communication technology at tertiary institutions should be enhanced with training opportunities for staff and faculty members. Training should include technical aspects of the technology as well as behavioural aspects, such as appropriate online

behaviour and suitability of different modes of communication for different tasks.

### **Suggestions for Further Research**

Faculty behaviour has shifted in terms of the amount of work that is being done from locations other than their office, such as at home, and that this shift is facilitated to the increased use and flexibility of electronic communication technology. This change could result in less interaction amongst colleagues, a condition which is seen as an essential part of faculty productivity (Baldwin, 1998; Cohen, 1996; Di Petta, 1998). Changes in face-to-face contact with students may also occur and such communication was viewed by some faculty as essential to their personal definition of productivity. This shift could be an interesting topic to explore in a future study, looking at possible adjustments in workload and altered patterns of contact between faculty and colleagues, faculty and administrators, and faculty and students.

While the study charted the amount of time spent at home on electronic communication it did not ask faculty members to report on the amount of time they spend each day physically at the office. It is possible that time spent at the office varies depending upon the school and the discipline, as well as individual preferences. A true comparison of School of Business faculty to other faculty would have included this question and future studies could perhaps investigate this issue more fully.

The appearance that women receive more electronic communications than their male counterparts merits further exploration. What is causing these additional e-mails: are they from students or from other faculty or from other parties? If they are from students, what prompts students to send more messages to female faculty than to male faculty? If they are not from students, who are they from and what, again, generates more electronic communications for this group?

As mentioned, use of electronic communication technology is evolving and changing. This technology has improved significantly since inception and users are becoming more experienced and sophisticated. A follow-up study in five or ten years' time may be interesting as an exploration of the progression of these elements. Items that faculty identified in this study as relevant, such as the need for timely feedback or the volume of e-mails, may no longer be significant, and other issues may emerge as topical. A follow-up interview with participants from this study would also be intriguing. Participants could implement some of their own or others' suggestions and a follow-up study could investigate the impact, if any, of these changes upon performance.

## **CHAPTER 7: RESEARCHER'S REFLECTIONS**

I started my journey into this topic because I was bothered by a sense of unknowingly letting students down — electronic communications could be sent to me late at night or early in the morning and, unless I was constantly checking my computer, I was not responding quickly to their questions or comments. While I appreciated the benefits that technology afforded me, I felt constantly behind in my work due to the unread e-mails in my inbox. I was curious to know if other faculty members were equally bothered and what they were doing to deal with this issue.

I learned a lot from the interviews and have changed my behaviour in terms of electronic communication. I realized that I needed to set some boundaries regarding accessibility, and so I included a written policy on my course outlines, indicating that students can expect a response within 24 hours during the week and not on weekends.

I am more sensitive to the efficacy of electronic communication for certain discussions. When I see strings of electronic communications going back and forth between individuals on one topic, I suggest a meeting or telephone conversation may be more productive.

I am more disciplined in my use of electronic communication, and consequently am spending less time on the system and reaping greater returns to my productivity. I am more aware of the individual response times that are out

there – not everyone takes the same approach to electronic communication that I do. Messages sent to one colleague will be answered immediately while those to another colleague will sit unread in their inbox for a few days.

I am much more at peace with the unread e-mails in my inbox – I now view my inbox as similar to the newspaper – not all of it will be read but, just as I don't feel the need to rip apart the paper to only include articles I will read, I don't feel the need to get rid of the items in my inbox.

I have had many months to ponder and muse on this issue of electronic technology and I hope that some of the insights that have emerged can be communicated to others and perhaps inform future behaviours.

### REFERENCES

- Allison, Genevieve J. (2004) Perceived and reported occupational stressors and coping strategies of selected community college business faculty members in Texas. Ph.D. dissertation, Texas A&M University, United States -- Texas. Retrieved March 5, 2008, from ProQuest Digital Dissertations database. (Publication No. AAT 3141384).
- Amey, M., & VanDerLinden, K. (2003). The use of technology: Institutional issues. *The NEA 2003 Almanac of Higher Education*, 11, 85-95.
- Antony, J. S., & Raveling, J. S. (1998). A comparative analysis of tenure and faculty productivity: Moving beyond traditional approaches. Paper presented at the Annual Meeting of the Association for the Study of Higher Education (23rd, Miami, FL, November 5-8, 1998). 42 pp.
- Atamian, R., & DeMoville, W. (1998). Office hours--none: An e-mail experiment. *College Teaching, 46*(1), 31.
- Baldwin, R. G. (1998). Technology's impact on faculty life and work. *New Directions for Teaching and Learning*, 76, 7.
- Beam, R. A., Kim, E., & Voakes, P.S. (2003). Technology-induced stressors, job satisfaction and workplace exhaustion among journalism and mass communication faculty. *Journalism and Mass Communication Educator*, *57*(4), 335.
- Bock, M. T. (1997). The measure of professorial productivity: Using student learning outcomes criteria. Paper presented at Annual Meeting of the Association for the Study of Higher Education (22nd, Albuquerque, NM, November 6-9, 1997). 31 pp.
- Brown, A. H., Benson, B., & Uhde, A. P. (2004). You're doing what with technology? An expose on "Jane Doe" college professor. *College Teaching*, *52*(3), p.100-104, Sum 2004.

- Camosun College (2003a). *Acceptable Computer Use*. Retrieved Feb. 12, 2008, from <a href="http://camosun.ca/about/policies/operations/o-1-college-systems-and-resources/o-1.3.pdf">http://camosun.ca/about/policies/operations/o-1-college-systems-and-resources/o-1.3.pdf</a>
- Camosun College (2003b). *E-mail Use*. Retrieved Feb. 12, 2008, from <a href="http://camosun.ca/about/policies/operations/o-1-college-systems-and-resources/o-1.4.pdf">http://camosun.ca/about/policies/operations/o-1-college-systems-and-resources/o-1.4.pdf</a>
- Camosun College (2007). Calendar, 2007-2008. (Available at www.camosun.ca).
- Camosun College (2008) *Camosun College Faculty Association*. Retrieved Feb. 21, 2008, from <a href="http://www.camosunfaculty.ca">http://www.camosunfaculty.ca</a>
- Carnavale, D. (2006). E-mail is for old people. *The Chronicle of Higher Education*, *53*, A27).
- Chen, C. (2001). Making e-mail requests to professors: Taiwanese vs. American students. Paper presented at the Annual Meeting of the American Association for Applied Linguistics (St. Louis, MO, February 24-27, 2001) 30 pp.
- Cohen, J. (1996). Computer mediated communication and publication productivity among faculty. *Internet Research*, 6(2-3), 41.
- Creswell, J. W. (2003). Research design: Qualitative, quantitative, and mixed methods approaches (2nd ed.). Thousand Oaks: Sage Publications.
- Daft, R., & Lengel, R. (1986). Organizational information requirements, media richness and structural design. *Management Science*, *32*(5), 554-571.
- Daft, R., Lengel, R., & Trevino, L. (1987). Message equivocality, media selection, and manager performance: Implications for information systems. *MIS Quarterly*, 13.
- Davidson, R. A. (2007, April). The virtue of restraint in virtual education; Or, why I refuse student e-mails. *The Teaching Professor, 21,* 2.
- Di Petta, T. (1998). Community on-line: New professional environments for higher education. *New Directions for Teaching and Learning*, *76*, 53.

- Duran, R., Kelly, L., & Keaten, J. (2005). College faculty use and perceptions of electronic mail to communicate with students. *Communication Quarterly*, 53(2), 159-176.
- Fisher, D., & Rubenson, K. (1998). The changing political economy: The private and public lives of Canadian universities. In J. Currie & J. Newson (Eds.), *Universities and Globalization: Critical Perspectives* (pp.77-98). Thousand Oaks, CA: Sage Publications.
- Flanagin, A. J. (2005). IM online: Instant messaging use among college students. Communication Research Reports, 22(3),13.
- Gall, M., Gall, J. P., & Borg, W. (2003). Educational research an introduction (7th ed.). Boston: Allyn and Bacon.
- Giddens, A. (1984). *The constitution of society*. Berkeley, CA: University of California Press.
- Giddens, A. (1991). Modernity and self-identity. Cambridge, U.K.: Polity Press.
- Giddens, A. (Ed.). (1993). *The Giddens reader*. Stanford, CA: Stanford University Press.
- Haworth, B. (1999). An analysis of the determinants of student e-mail use. Journal of Education for Business, 75(1), 55.
- Hilton, J. (2006). The future for higher education: Sunrise or perfect storm? *EDUCAUSE Review, 41*(2), 59.
- Huber, M. T. (1998). Community college faculty attitudes and trends, 1997. Stanford, CA: National Center for Postsecondary Improvement.
- Ishii, K. (2005). The human side of the digital divide: Media experience as the border of communication satisfaction with e-mail. *Journal of Technical Writing & Communication*, *35*(4), 385-402.
- Katula, R. A., & Doody, A. (1990). The collegiality model: An alternative for evaluating faculty productivity. *Association for Communication Administration Bulletin, 74*, 74-82.

- Kelly, L., Keaten, J. A., & Finch, C. (2004). Reticent and non-reticent college students' preferred communication channels for interacting with faculty. *Communication Research Reports*, *21*(2), 197-209.
- Krause, K., Hartley, R., James, R., & McInnis, C. (2005). The first year experience in Australian universities: Findings from a decade of national studies. *Centre for the Study of Higher Education, University of Melbourne*, 1-115. (Available at http://www.cshe.unimelb.edu.au/pdfs/FYEReport05KLK.pdf).
- Lee, A. (2006). Survey of School of Business Students, Camosun College. (Available from the School of Business, Camosun College, 4461 Interurban Road, Victoria, B.C. V9E 2C1).
- Levin, H. M. (1991). Raising productivity in higher education. *Journal of Higher Education*, *62*(3), 21.
- MacAskill, J. (Ed.). (2006a). About Camosun: Camosun College A life-changing experience. Retrieved June 8, 2007 from http://www.camosun.bc.ca/about/index.html
- MacAskill, J. (Ed.). (2006b). Camosun's vision, mission & values. Retrieved June 8, 2007 http://www.camosun.bc.ca/about/vision-mission-values.html
- Macintosh, N. B. (1994). Management accounting and control systems An organizational and behavioral approach. West Sussex, U.K.: John Wiley & Sons.
- Marshall, C., & Rossman, G. B. (2006). Designing qualitative research.

  Thousand Oaks, CA: Sage Publications.
- Massy, W. F., & Wilger, A. K. (1995). Improving productivity: What faculty think about it -- and its effect on quality. *Change, 27*(4),10.
- McBeath, G. (1974). Productivity through people a practical guide to improvement. London, UK: Business Books.
- McInnis, C. (2002). The impact of technology on faculty performance and its evaluation. *New Directions for Institutional Research*, 114, 53.

- Middaugh, M. F. (2001). *Understanding faculty productivity: Standards and benchmarks for colleges and universities*. San Francisco, CA: Jossey-Bass Publishers.
- Montano, B. R., & Dillon, R. (2005). The impact of technology on relationships within organizations. *Information Technology and Management, 6(2-3),* 25.
- Morgan, G. (1998). *Images of organization The executive edition*. San Francisco: Berrett-Koehler Publishers, Inc.
- Ogan, C., & Chung, D. (2003). Stressed out! A national study of women and men journalism and mass communication faculty, their uses of technology, and levels of professional and personal stress. *Journalism and Mass Communication Educator*, *57*(4), 352.
- Patton, M. Q. (1987). How to use qualitative methods in evaluation. Newbury Park, CA: Sage Publications.
- Pletka, B. (2007). Educating the net generation: How to engage students in the 21st century. Santa Monica, CA: Santa Monica Press.
- Polster, C., & Newson, J. (1998). Don't count your blessings: The social accomplishments of performance indicators. In J. Currie & J. Newson (Eds.), *Universities and globalization: Critical perspectives*. Thousand Oaks, CA: Sage Publications.
- Powell, W. W., & Snellman, K. (2004). The knowledge economy. *Annual Review of Sociology, 30*(1), 199-220.
- Ranson, K. A. (2008) *Mark Zuckerberg*. In Britannica Book of the Year, 2008. Retrieved March 5, 2008, from Encyclopædia Britannica Online: http://search.eb.com.proxy.lib.sfu.ca/eb/article-9439426
- Ray, M. (2008). Social Networking—Making Connections on the Web. In Britannica Book of the Year, 2008. Retrieved March 5, 2008, from Encyclopædia Britannica Online: http://search.eb.com.proxy.lib.sfu.ca/eb/article-9439169

- Roberson, J. (2004). An analysis of email as a communications tool in the organization workplace environment. PhD. dissertation, Fielding Graduate Institute, United Sates California. Retrieved March 5, 2008, from ProQuest Digital Dissertations database. (Publication No. AAT 3132467).
- Roethlisberger, F. J., & Dickson, W. J. (1939). *Management and the worker*. Cambridge, MA: Harvard University Press.
- Royal Roads University. (2007). *Netiquette*. Retrieved Feb. 12, 2008, from <a href="http://learn.royalroads.ca/ctet/academic/Resources/netiquette.htm">http://learn.royalroads.ca/ctet/academic/Resources/netiquette.htm</a>
- Shepherd, S. (2004). Relationships between computer skills and technostress:

  How does this affect me? Myrtle Beach, CA: Association of Small
  Computer Users in Education Conference. (37<sup>th</sup>, Myrtle Beach SC, June 6-10, 2004) 7 pp.
- Shields, M. (2007). Facebook sales 'micro' managed. *MediaWeek*, 17(9), 7.
- Skovholt, K., & Svennevig, J. (2006). E-mail copies in workplace interaction. *Journal of Computer-Mediated Communication*, 12(1), 42-65.
- Stake, R. E. (2005). Qualitative Case Studies. In N. K. Denzin & Y.S. Lincoln (Eds.), *The Sage Handbook of Qualitative Research* (3rd ed.) (pp. 443-466). Thousand Oaks, CA: Sage Publications.
- Tabers, K. E. (2002). Information technology faculty in the community college: Perspectives on change at the speed of light. PhD. dissertation, Loyola University of Chicago, United States - Illinois. Retrieved March 5, 2008, from ProQuest Digital Dissertations database. (Publication No. AAT 3039309).
- Tapscott, D. (1998). *Growing up digital: The rise of the net generation*. New York: McGraw-Hill.
- Thomas, S. L. (2006). Sampling rationale and rigor in choosing what to observe. In C. F. S. Conrad, (Ed.), *The Sage Handbook for Research in Education* (pp. 393-404). Thousand Oaks, CA: Sage Publications.

- Tierney, W. G. (1999). Building the responsive campus: Creating high performance colleges and universities. Thousand Oaks, CA: Sage Publications.
- Tinis, M. (2006). Camosun College community report. Retrieved June 8, 2007, from http://www.camosun.bc.ca/community-report06.pdf
- Toma, J. D. (2006). Approaching rigor in applied qualitative research. In C. F. Conrad & R.C. Serlin (Eds.), *The Sage Handbook for Research in Education* (pp. 405-424). Thousand Oaks, CA: Sage Publications.
- Tytherleigh, M. Y., Webb, C., Cooper, C. L., & Ricketts, C. (2005). Occupational stress in UK higher education institutions: A comparative study of all staff categories. *Higher Education Research and Development, 24*(1), 41.
- Waldvogel, J. (2007). Greetings and closings in workplace e-mail. *Journal of Computer-Mediated Communication*, 12(2), 122-143.
- Wang, Y. & Cohen, A. (1998). University faculty use of the Internet. *National Convention of the Association for Educational Communications and Technology (AECT)*. 455-462.
- Webster, F. (2002). Theories of the information society (2nd ed.). London: Routledge.
- Yin, R. K. (1994). *Case study research: Design and methods* (2nd ed.). Thousand Oaks, CA: Sage Publications.

## **Appendix A: Recruitment letter to School of Business Faculty**

August 15, 2007

Dear Colleagues,

As some of you may be aware, I am engaged in a doctoral program at Simon Fraser University. As part of this program I will be conducting research in September and October 2007 on the impact of electronic communication technology on faculty productivity. I have approval from the College's Ethics committee to conduct this research.

The purpose of this letter is to recruit volunteers for my research. Participants must be full-time faculty within the School of Business whose primary job is instruction – that eliminates both Chairs and Program Leaders.

Participants will complete a time log and will then be interviewed by me on an individual basis. The interview will take about one hour. Completion of the time log will require you to track and record your electronic communication usage for a one week period and will take about five minutes each day. I will be selecting participants from those that volunteer who exhibit a range of characteristics. Participants may withdraw from the study at any time. The data of this study will maintain confidentiality of your name and the contributions you have made to the extent allowed by law.

If you are interested in participating, or have questions for me, please complete the bottom of this letter and return it to my mailbox.

inanks,			
Sheila Elworthy			
Name:			
I have some questions. Please ca	all me.		
I am interested in participating.			
My age is in the range of (please circle)	25 – 40	41-55	over 55
Number of years teaching at the post-sec	ondary level _		
Discipline of instruction (ie. Accounting, E	conomics, etc.	)	

## Appendix B: Time Log

October 1, 2007

Dear Participant,

Thank you for agreeing to participate in the research component of my doctoral program in Educational Leadership which I am completing at Simon Fraser University. My research is studying the impact of electronic communication on faculty productivity, with faculty at Camosun College being my research population. Your involvement includes completion of a time log and participation in an interview. I will schedule the interview with you upon completion of the time log. I have approval from the College's Ethics committee to conduct this research.

Please be assured that this data will be kept confidential. The data will be correlated and results will not be attributed to any individual by name in the study. Please call me if you have **any questions at all** while completing this log.

(889-2994 cell, 592-2742 home, 370-4136 office)

There are three parts to this time log.

In **Part 1** please put one mark for every electronic communication that you send today that was self-generated as opposed to being in direct response to an electronic communication that you receive.

Part 2 is separated into your responses to electronic communications received today and electronic communications received on prior days. In both sections, please put one mark in the chart for each electronic communication related to your work as a faculty member at Camosun College dealt with each day for seven consecutive days. For each electronic communication (e-mail, text message, etc.) please decide how relevant it is to your work (columns) and what action you took (rows).

In Part 3, please record how many minutes you spend on Camosun College work-related electronic communication during the day and the number of times during the day that you check your electronic communication devices, which could include e-mail, websites and blackberries. Please record only the time spent on communication; if you are adding course material to your website, for example, that time would not be included but if you are posting a message to your website that time would be included. Please do not include telephone use or facsimile use. Include only text messaging and e-mail use of blackberries and cell phones.

Please return these logs to me as soon as the week is completed. Again, do not hesitate to call me should you have any questions about the completion of this log or my study in general.

Thank you again for your time,

Sheila

# Appendix B (cont'd): Time Log

Tuesday,	Oct.	9/07
Part 1		

Participant # 37859

Number of work-related electronic communications
sent today (these are self-generated, so not directly
in response to an incoming communication)
5 10

F	•	<u>a</u>	ſ	t	<u>2</u>	

Work-related communications received today	Very relevant to my work	Somewhat relevant to my work	Unsure of relevance to my work	Not relevant to my work
Deleted without reading entire body of message				
Deleted after reading				
Replied and/or forwarded				
Other action (phoned, met with person, etc.)				
Took no action (didn't read body of message, didn't delete)				

Work-related communications received before today	Very relevant to my work	Somewhat relevant to my work	Unsure of relevance to my work	Not relevant to my work
Deleted without reading entire body of message				
Deleted after reading				
Replied and/or forwarded				
Other action (phoned, met with person, etc.)				
Took no action (didn't read body of message, didn't delete)				

Part 3	From my work-site	From a remote location
Total minutes spent on work-related electronic communication		
Number of times I checked my electronic communication devices today		

	Date
Housel	keeping issues:
•	Please read and sign the consent form – any questions?
•	I will send you a copy of the transcript in a few weeks which I would like you to read to ensure that the transcription is accurate. Are you willing to do that?
•	Please note that when I talk about electronic communication I am referring to your use of electronic communication technology for doing your work — interactions with students, with other faculty, administration, textbook publishers, etc. I am not talking about communication to your friends.
•	Please note that when I say electronic communication I am not talking about telephones or facsimiles. I am talking about e-mail, text messaging, etc.
•	Some of my questions will be specific to student contact and will say electronic communication with students.
•	Feel free to say "I don't know" if that is the case – some of the questions ask for detailed information that may not be relevant to you!
•	This interview should take about one hour.
•	Any questions about that or about the interview itself?
	ow long have you been teaching at Camosun College? How long in total ave you been teaching?
rai	hich School and which discipline do you primarily teach in? Which age nge are you within? Do you have internet access at home? Is it high-eed internet?
stı	hat modes of communication are you using for communicating with your udents outside of the classroom? (Website postings, blogging, e-mail, face-face, telephone, msn)
co	hat percentage of your working hours is spent communicating via electronic mmunication (exclude telephone and fax) with students? (0-20%, 20-40%, 0-60%, more than 60%)
	utside of class time, what percentage of your communication with students, via: Electronic communication technologyFace-to-faceTelephoneOther
	uring an average week, how much time in total do you spend ommunicating with students outside of class time via all media?

7.

more than 60%)

What percentage of your working hours is spent communicating via electronic communication (exclude telephone and fax) with others, such as administrators, publishers, support staff, etc.? (0-20%, 20-40%, 40-60%,

### Appendix C (cont'd): Interview questions

- 8. What percentage of your communication with others, is via:

  Electronic communication technology \_\_\_\_Face-to-face \_\_\_\_Telephone \_\_\_\_
  Other \_\_\_\_
- 9. During an average week, how much time in total do you spend communicating with administrators, publishers, support staff via all media?
- 10. Do you save e-mails? If so, what percentage of e-mails do you save? For how long? Why do you save them?
- 11. Do you print e-mails? If so, what percentage of e-mails you receive do you print? Why do you print them?
- 12. Has the amount of time you spend on electronic communication changed as compared to one year ago?
- 13. Are there periods throughout the semester when time spent on electronic communication increases or decreases? What factors might be causing this change?
- 14. What is the nature of the student electronic communications? (Asking for assistance on course content, asking for assistance on assignments, administrative (due date, sickness))
- 15. Does electronic communication contact with students impact student use of your office hours? If so, how?
- 16. What do you consider to be an appropriate response time to student electronic communication requests?
- 17. How quickly do you tend to respond to student electronic communication requests?
- 18. Do you have any policy on electronic communication in your course outline or that you discuss with students at the start of a semester?
- 19. Do you check electronic communication on week-ends? How often do you check? Do you respond to electronic communications on weekends?
- 20. Do you check electronic communication in the evening? How often do you check? Do you respond to electronic communications in the evening?
- 21. Do you check electronic communication when you are on holiday? How often do you check? Do you respond to electronic communications when on holiday?

#### Appendix C (cont'd): Interview questions

- 22. How, if at all, is electronic communication with 'English as a Second Language' students different from electronic communication with other students?
- 23. How, if at all, is electronic communication with 'older' students different from electronic communication with other students?
- 24. In terms of work, what typically do you use electronic communication for? In terms of work, what, typically, don't you use electronic communication for?
- 25. How would you personally define productivity with regard to your work as a faculty member?
- 26. Are there certain aspects of electronic communication technology that increase your productivity? If so, please explain.
- 27. Are there certain aspects of electronic communication technology that decrease your productivity? If so, please explain.
- 28. What strategies have you found helpful for improving your productivity regarding use of electronic communication devices?
- 29. If you could change anything about electronic communications, what change would you make that might increase your productivity? What could be done to improve use of this technology at the College?
- 30. Overall, do you think electronic communication technology makes you more or less productive? Why? Is there anything else I should know about the connection between electronic communication technology and your productivity?
- 31. Did you discover anything while completing the time logs? If so, what?
- 32. Did you change your pattern of dealing with electronic communication while completing the time log? If so, how?
- 33. I notice that many people did not deal with old e-mails on a daily basis. What happens to your old e-mails? How many e-mails do you have in your inbox right now? How many are from yesterday or earlier? Is that the norm for you?

Please feel free to contact me should you have any other thoughts on this topic. As well, I will send you a copy of the transcript from this interview so that you can verify the transcription. Thank you for your time.

### **Appendix D: Informed Consent Form**

Electronic Communication and Faculty Productivity Research Project Informed Consent

I understand that the purpose of this study is to assess the impact of electronic communication technology on faculty productivity at the college. This study includes email and other electronic communication devices that faculty might be using, such as blackberries and the text messaging capability of cellular phones, but does not include telephones and facsimile machines.

I understand that the researcher may make presentations to other educators regarding the outcomes of this study and that these presentations may also include writing articles for professional educational journals.

I understand that the data results may be used by the researcher in future research projects that involve her and possibly other researchers.

I understand that my participation involves a one-on-one interview with the researcher. My participation also involves completion of a time log for a one or two week period occurring between September 5, 2007 and November 30, 2007.

I understand that all my comments, written or oral will remain completely anonymous to those receiving the presentations, reports and/or reading the articles related to the project.

I understand that I will be given the opportunity to review the interview transcription to ensure that the transcription reflects my opinions and comments.

I understand that knowledge of my involvement in this project will be kept confidential by the researcher.

I understand that participation in this project is voluntary and that no coercion has been used to obtain my cooperation.

I understand that I may withdraw from the project at any time and that knowledge of my withdrawal will be confidential.

I understand that participation in this project is independent of and not connected to my employment at the college and has been approved by the Ethics boards of both Camosun College and Simon Fraser University.

I wish to give my cooperation as a participant in this research project.

Signed: _	 	 * ***	 _
Date:	 	 · · · · · · · · · · · · · · · · · · ·	
Place:	 	 	

## **Appendix E: List of Codes**

- 1. Expected response times
- 2. Actual response times
- 3. Frequency of checking
- 4. Electronic communication is good for
- 5. Electronic communication is not good for
- 6. English as a Second Language students
- 7. Older students
- 8. Strategies used to improve productivity
- 9. Suggested changes
- 10. Ways in which productivity increases and decreases
- 11. Dealing with old e-mail
- 12. Verbal and written policies
- 13. Printing and saving behaviours

# Appendix F: Tabular results of Interviews and Time logs

Table F1. Schools within which participants teach

	All faculty (16)
School of Business faculty	9
Arts and Sciences	4
Trades & Technology	2
Health & Human Services	1

Table F2. Number of years teaching at Camosun College

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Range	1 to 25 years	1 to 25 years	2 to 13 years
Average	8 years	8 years	7 years

Table F3. Number of years teaching in total

	All faculty (16)	SOB faculty (9)	Other faculty (7)
Range	2 to 27 years	2 to 27 years	4 to 17 years
Average	10 years	10 years	10 years

Table F4. Age range of participants

	All faculty (16)	SOB faculty (9)	Other faculty (7)
21 to 30 years	2	0	2
31 to 40 years	2	1	1
41 to 50 years	5	4	1
51 to 60 years	7	4	3

## Appendix F (cont'd): Tabular results of Interviews and Time logs

Table F5. Time log results for e-mails received during the work week

	Very relevant to my work	Somewhat relevant to my work	Unsure of relevance	Not relevant to my work	Total	
School of Business facult	y (9):					
Deleted without reading entire body of message	2	4	27	36	69	
Deleted after reading	65	49	9	22	145	
Replied and/or forwarded	98	25		20	143	
Other action (phoned, met with person, etc.)	22	3	1	1	27	
Took no action (didn't read, didn't delete)	7	9	2	3	21	
Total	194	90	39	82	405	
Average number of e- mails received per person per day					9	
Other faculty (7):						
Deleted without reading entire body of message		18	25	74	117	
Deleted after reading	68	54	11	18	151	
Replied and/or forwarded	122	56	15	13	206	
Other action (phoned, met with person, etc.)	17	8	1		26	
Took no action (didn't read, didn't delete)	2	13	7	2	24	
Total	209	149	59	107	524	
Average number of e- mails received per person per day					15	
A7-7- B1 1						

Note. Numbers represent the volume of e-mails received by all participants within each group from Monday to Friday.

### Appendix F (cont'd): Tabular results of Interviews and Time logs

Table F6: Handling of old work-related e-mails during the work week

	T.,					
	Very relevant to my work	Somewhat relevant to my work	Unsure of relevance	Not relevant to my work	Total	
School of Business facult	y (9):					
Deleted without reading entire body of message	2	1	3	12	18	
Deleted after reading	29	16	3	2	50	
Replied and/or forwarded	24	3		4	31	
Other action (phoned, met with person, etc.)	10	2			12	
Took no action (didn't read body of message, didn't delete)	4		1	1	6	
Total	69	22	7	19	117	
Other faculty (7):		·				
Deleted without reading entire body of message	7	1 .	3	4	15	
Deleted after reading	6	5	5	1	17	
Replied and/or forwarded	13	17	1	1	32	
Other action (phoned, met with person, etc.)	8	1			9	
Took no action (didn't read body of message, didn't delete)	1	4	1		6	
Total	35	28	10	6	79	

Note. This table summarizes the handling of old work-related e-mails from Monday to Friday by the two faculty groupings. Old e-mails are ones received on a day prior to the day in question.